Import Competition, Labour Market Regulations, and Firm Outsourcing^{*}

Pavel Chakraborty[†]

Asha Sundaram ‡

March 2020

Abstract

Using unique information on firm level outsourcing of manufacturing jobs by Indian firms, we propose two novel channels as determinants of the fragmentation of production: import competition and labour market regulation. We find that greater import competition from China is associated with a significant increase in outsourcing – a 10 percentage point increase in the import penetration ratio leads to a 0.24-0.50 percentage point increase in the share of outsourcing expenses in total expenses of a firm. This effect is mitigated for firms operating in states with pro-employer labour laws by 0.17-0.24 percentage points. These results are primarily driven by domestic, multi-product firms producing final goods. Outsourcing firms have lower marginal cost, charge a lower price, and expand in response to import competition. In addition, we find a corresponding increase in (a) the likelihood that informal sector firms engage in an outsourcing contract with other firms, and (b) output per worker in the informal sector, where labour laws are not enforced.

JEL classifications: F1, F12, F14, F16, J08, J46

Keywords: Chinese Import Competition, Outsourcing of Manufacturing Jobs, Inflexible Labour Laws, Informal Sector, Costs, Firm Organization

^{*}We thank Reshad Ahsan, Andrea Ciani, Stephanie Haller, Davide Rigo, Tara Shankar Shaw, Chiara Tomassi, Maurizio Zanardi and conference participants at 10th Conference on Economics of Global Interactions, Bari; 14th ISGEP Workshop, Trento; Midwest Trade Meetings, Fall 2019, St Louis and seminar participants at the Center for Advanced Financial Research and Learning (CAFRAL), Reserve Bank of India (RBI) and Indian Institute of Technology, Mumbai for their valuable comments. This paper was partly written while Sundaram was a research fellow at CAFRAL, RBI. The usual disclaimer applies.

[†]Department of Economics, Management School, Lancaster University, LA1 4YX, UK. Email: p.chakraborty1@lancaster.ac.uk

[‡]Department of Economics, University of Auckland, Auckland 1010, NZ; Email: a.sundaram@auckland.ac.nz

1 Introduction

"We live in an age of outsourcing" (Grossman and Helpman, 2005).

Fragmentation of production activity has received extensive attention in the literature in recent years. According to Grossman and Helpman (2005), firms now subcontract or outsource a range of activities – jobs related to both manufacturing (such as product design, assembly, research and development) and professional services (marketing, distribution, after-sales service).¹ Feenstra (1998) cites the production of Barbie dolls as an example to highlight the increase in foreign outsourcing as a result of the spectacular integration of the global economy during the 80s and 90s.² Ever since, studies have examined several possible determinants of vertical integration by firms ³. These include the potential for holdup problems (Grossman and Helpman, 2002, 2005; Ornelas and Turnner, 2008), product market competition (Aghion et al., 2006; Hortacsu and Syverson, 2007; McGowan, 2017), prices (Legros and Newman, 2013; Alfaro et al., 2016), offshoring (Bernard et al., 2018), contractibility (Grossman and Helpman, 2005; Alfaro et al., 2019), trade/globalization (McLaren, 2000; Chen et al., 2004; Ornelas and Turnner, 2011; Chongvilaivan and Hur, 2012; Buehler and Burghardt, 2015; Stiebale and Vencappa, 2018), and communication technology (Fort, 2017).⁴

In this paper, we propose two new channels that can influence outsourcing of production activity of a firm outside it's boundary: (i) international trade; particularly, import competition from China in the domestic market,⁵ (ii) labour market regulation. From a theoretical point of view, the impact of competition on vertical integration⁶ is ambiguous and may be non-monotonic (Legros and Newman, 2014). Higher degree of product market competition may increase (Ornelas and Turner, 2008, 2011) or decrease (McLaren, 2000, 2003; Grossman and Helpman, 2002, 2005; Buehler and

¹The Annual report of the World Trade Organization (1998) details the production of a particular "American" car as follows: "30% of the car's value goes to Korea for assembly, 17.5% to Japan for components and advanced technology, 7.5% to Germany for design, 4% to Taiwan and Singapore for minor parts, 2.5% to the United Kingdom for advertising and marketing services and 1.5% to Ireland and Barbados for data processing. This means that only 37% of the production value is generated in the United States (p. 36)."

²According to Feenstra, Mattel procures raw materials (plastic and hair) from Taiwan and Japan, conducts assembly in Indonesia and Malaysia, buys the moulds in the U.S., clothing in China and paints used in decorating the dolls in the U.S.

 $^{^{3}}$ A related literature examines the impacts of a firm's offshoring decision. This set of studies looks at how the offshoring decision affects a range of variables, including skill premia (Feenstra and Hanson, 1999), shares of skilled workers within a firm (Mion and Zhu, 2013), wages and employment (Hummels et al., 2014), polarization (Harrigan et al., 2016) and trade (Bernard et al., 2018).

⁴Most of these studies are theoretical in nature, with a few exceptions.

⁵Rodriguez-Lopez (2014) proposes a theoretical model that looks at how import competition and the interaction between import competition and productivity of firms can lead firms to offshore.

⁶Vertical integration is the opposite of outsourcing.

Burghardt, 2015; McGowan, 2017) incentives to vertically integrate. Aghion et al. (2006) predict a U-shaped relationship between competition and vertical integration.⁷ In marked contrast to the impressive body of theoretical work on the link between trade and outsourcing, the empirical evidence is scant. Additionally, to our knowledge, no study has proposed labour regulation as a channel through which trade impacts outsourcing. We submit that labour regulation that increases the cost of employing labour in-house can incentivize firms to outsource to avail of a lower marginal cost of production, particularly in the face of greater import competition.

A key contribution of our work is that we bring to bear unique data on outsourcing that, we believe, appropriately captures outsourcing activity at the firm level. Previous studies analyzing the organization of firms capture vertical integration using industry level input-output tables to calculate the proportion of inputs into production produced within a firm (Acemoglu et al., 2009, 2010; Alfaro et al., 2016; Stiebale and Vencappa, 2018; Liu et al., 2019), except for Fort (2017).⁸ Such industry level information is subject to caveats. First, the international trade literature has documented substantial heterogeneity across firms within industries on the composition and quality of inputs used in production. Industry level input-output tables may fail to capture this heterogeneity (DeLoecker et al., 2016). Second, firms may both produce and outsource production of the same input, as noted by Bernard et al. (2018). We argue that our measure of firm outsourcing activity overcomes these concerns by directly exploiting data on outsourcing expenditure by firms explicitly on manufacturing jobs.⁹ Our variable, defined as the share of expenses on outsourcing of manufacturing jobs in total firm expenses, captures all expenses incurred by firms to have their manufacturing requirements fulfilled from outside parties.¹⁰ We argue that it is closest in spirit to Grossman and Helpman's (2005) definition of outsourcing: it is more than just the purchase of raw materials and intermediate goods. It indicates a bilateral relationship(s), where the partner makes a relationship-specific investment to produce goods that fit the firm's particular needs.

Panel A of **Figure 1** presents the average share of expenditure on outsourcing of manufacturing

⁷According to their theory, a small increase in competition reduces producers' incentives to integrate by raising suppliers' investment incentives. For high degrees of competition, the model predicts that producers will have a high incentive to vertically integrate because independent suppliers can capture most of the surplus generated by the producer.

⁸Fort (2017) is the only study to use data on fragmentation of economic activity at the firm level. However, the information used is only for two years, 2002 and 2007, and based on a survey. We use a panel where we observe the pattern of outsourcing at the firm level over a significant period of time. However, unlike Fort (2017), we cannot distinguish between international and domestic outsourcing.

⁹The dataset also provides information on outsourcing of professional services, but we use that variable as a placebo. More details in Section 4.2.

¹⁰Our primary measure of outsourcing captures the intensive margin of outsourcing. We also look at the percentage of firms involved in outsourcing, or the extensive margin. Results remain qualitatively similar.

activities by firms before and after 2001. The break in 2001 is intended to capture the impact of China's assession to the WTO.¹¹ An average firm spent about 0.15% of its total expenses on outsourcing between 1995 and 2001, which shot up to 1% between 2002-2007; a jump of about 5.6 times or an increase of 560%. **Panel B** looks at how the incidence of outsourcing activity has changed over time. It shows that in a year the average percentage of firms involved in outsourcing was around 8% between 1995 and 2001, which increased to about 27% between 2002 and 2007, an increase of about 350%.

To understand whether international trade is one of the main forces in driving this observed significant change in the way firms organize production, we exploit the increase in import competition faced by Indian firms from China post China's accession to the WTO as a quasi-natural experiment.¹² We argue that using Chinese competition as a proxy for an exogenous increase in import competition is valid for the following reasons. First, China is currently India's largest trading partner. **Figure 2** plots Indian imports from China between 1995 and 2007. The share of manufacturing imports from China as a share of total manufacturing imports skyrocketed from less than 5% in 1995 to almost 25% in 2007 – an increase of 400%. The figure shows that this steep acceleration is particularly visible after China's accession to the WTO in 2001. We observe a similar pattern for the import penetration ratio, which increased from less than 1 to almost 8% in the same time period.¹³

Table 1 compares India's trade with China and other large trading partners at three different points in time: 1992, 2001 and 2007. It shows that China accounted for the largest increase in India's imports relative to other countries and major regions of the world. India's share of Chinese imports grew by around 9000% between 1992 and 2007.¹⁴ In comparison, imports from

¹¹China's membership to the WTO in 2001 was one of the most important episodes of world trade in the last two decades. China's export performance post-1990, and more so since 2001, has been spectacular. Its exports grew from US\$ 62 billion to US\$ 1.2 trillion between 1990 and 2007; an average of around 20% per year (Iacovone et al., 2013). In the same period, China's share of GDP more than doubled, from 15.9 to 34.9%. Following this very strong export performance, China became the world's largest exporter in 2009, and the second largest economy in 2010 (Iacovone et al., 2013). Naturally, this meteoric rise of China to the status of a global exporting giant, particularly of manufactured goods, has prompted economists to examine the effects of import competition from low-wage countries, specifically China, on various firm and industry level outcomes in developed countries (Bernard et al., 2006; Liu, 2010; Autor et al., 2013; Mion and Zhu, 2013; Martin and Meajean, 2014; Bloom et al., 2016), and to a far lesser extent in developing countries (Iacavone et al., 2013 and Utar and Torres-Ruiz, 2013 for Mexico and Medina, 2017 for Peru).

 $^{^{12}}$ There is precedence in the literature to treat the sharp rise in China's share in total imports of countries (both developed and developing) due to its accession to the WTO in 2001 as a quasi-natural experiment (see, Lu and Yu, 2015; Bloom et al., 2016).

 $^{^{13}}$ The Chinese import penetration ratio is calculated as the share of Chinese imports in an industry in total domestic production, including imports and exports. See **Appendix A** for definitions of key variables.

¹⁴Note that the percentage increase in Chinese imports in the case of India is almost 9 times higher when compared to the US during the same time period; the percentage increase for the US was 1156 during 1991–2007 (Autor et al., 2013).

ASEAN (another large trading partner), the US and the EU increased by 888%, 230% and 132%, respectively. Compared to Mexico¹⁵, where the Chinese share of manufacturing imports increased by a factor of 8, in the case of India it increased by a factor greater than 90 over the same time period (1992–2007).¹⁶

Secondly, the growth in Chinese exports to India as a result of accession to the WTO is a result of China's internal reforms to a market-oriented economy. This transition to a market economy from a centrally planned economy resulted in significant productivity growth for Chinese firms, which was further bolstered by a reduction in trade costs as a result of its accession to the WTO. We treat this as a unilateral trade shock and not a mutual trade expansion.¹⁷ Given China's dominance in India's trade and the phenomenal increase in outsourcing activity by firms in the post-2001 period, our question of whether Indian manufacturing firms respond to import competition from China by increasing outsourcing is a pertinent one.

We examine the impact of import competition from China on outsouring differentially for firms located in states with pro-employer versus pro-worker labour regulation. India is a useful setting for this purpose. There is substantial heterogeneity in labour market regulation across Indian regions.¹⁸ Differences in labour laws across states provide ample variation to understand whether the gains to firms from outsourcing are particularly large when pro-worker labour laws act as a tax on employing labour in-house in the formal sector. Greater import competition may therefore be associated with more outsourcing in regimes with pro-worker labour laws, relative to pro-employer labour regimes. Focusing on a federal democracy like India allows us to delve into the role played by labour regulation in determining the relationship between trade and outsourcing.¹⁹ We follow

¹⁹In order to determine this relationship, a crucial identifying assumption must be met: Chinese import competition should be exogenous to the labour regime. In other words, it should not be caused by changes in outsourcing patterns

¹⁵A large number of studies exploring the impact of Chinese import competition on developing countries focus on Mexico (Iacovone et al., 2013; Utar and Torres-Ruiz, 2013).

¹⁶We present Chinese imports by India as a share of Indian imports from the world across manufacturing industries in **Table 15** (**Appendix B**). Imports from China are largest in labour-intensive industries like textiles and wood and in machinery and transport equipment.

¹⁷This approach requires that the import demand shock to India, especially after 2001, was not the primary cause of China's export surge. While it seems plausible that China's export growth to India during the 2000s was a result of supply shocks internal to China, we use imports from China by other developing countries (Brazil, Mexico, Indonesia and Malaysia) as an instrument for Chinese imports to India. All approaches yield similar results.

¹⁸Labour laws in India are guided by the Industrial Disputes Act of 1947 (hereafter IDA, 1947). The Act sets out the regulations governing employer-employee relations and the legal procedures to be followed in the case of labour disputes in the factory sector. The IDA was passed by the central government, but has been extensively amended by state governments causing Indian states to differ markedy in their labour laws. Besley and Burgess (2004) read all state level amendments made to the IDA during 1958-1995 in 16 major Indian states (from Malik, 1997). Each amendment is coded as being either pro-worker, neutral, or pro-employer, depending on whether it lowered, left unchanged or increased an employer's flexibility in hiring and firing factory workers, respectively. Based on the cumulative scores, they classified states as "pro-worker", "pro-employer", and "neutral". We discuss this in detail in Section 4.2.

Gupta et al. (2009) and Adhvaryu et al. $(2013)^{20}$ and exploit this variation in labour regimes across Indian states, noting that other institutional factors such as the monetary policy regime are fixed at the country level.

We have three sets of results. First, an increase in Chinese import penetration, particularly product market competition and not competition in intermediate input markets, significantly increases the share of expenses on outsourcing of manufacturing jobs by Indian manufacturing firms. A 10 percentage point increase in the Chinese import penetration ratio is associated with an increase in the outsourcing share of total expenses for an Indian manufacturing firm of 0.24–0.50 percentage points. The result is mainly driven by domestic multi-product firms producing final goods. The effect is significant for both large and small firms, but bigger for large firms.

Second, the increase in outsourcing is mitigated for firms in Indian states with pro-employer labour laws by 0.17-0.24 percentage points, suggesting that import competition increases outsourcing relatively more in states with pro-worker labour regulation. This finding is consistent with the idea that pro-worker labour laws magnify the positive relationship between import competition and outsourcing activity by acting as a tax on labour, increasing the costs of using in-house labour in the formal sector. For example, sticky wages in states with pro-worker labour laws can add to overall production cost of a firm if it wants to increase production. Or, if a firm wants to alter the size of its workforce in the short run in response to Chinese competition, labour laws may restrict it from doing so in the presence of hiring and firing regulations.²¹ We find that outsourcing firms located in states with pro-worker labour laws can lower marginal costs by 25% and prices by about 15% more than firms located in states with pro-employer labour laws in response to Chinese import competition. In addition, their sales increases by about 32% more than outsourcing firms located in pro-employer states.

All results are robust to controlling for a battery of industry and firm characteristics, industry level import tariffs, availability of cheaper intermediate inputs from China, export market competition, interactions between state and year fixed effects, and a host of other checks. We also conduct a placebo test, where we examine outsourcing of professional jobs (where labour regulations do not apply) as our outcome of interest. We find no evidence that labour regulation plays a role in

in the industrial sector across different labour regimes or by other factors that may affect outsourcing. We argue that the identifying assumption is met in our case; a large majority of labour Acts were enacted in the period 1949-1989. In the nineties, the legislative activity came to a halt, with no new amendments in the IDA (Ahsan and Pages, 2008).

 $^{^{20}}$ The classifications used in these papers are based on Besley and Burgess (2004) and its critique by Bhattacharjea (2006).

²¹Section V-B of the IDA 1947 lays out special provisions that apply only to industrial establishments employing at least one hundred workers. This section is more draconian - it requires that no workers may be laid-off or retrenched without the prior permission of the government.

determining the relationship between import competition and outsourcing of professional jobs.²²

Finally, we use data on outsourcing activity by manufacturing enterprises (micro) in the Indian informal sector to check for evidence on the symbiotic relationship between the formal and informal sectors. Like many developing economies, India has a large informal sector consisting of enterprises employing less than ten workers. Firms in the informal sector face lower labour costs because labour laws are not enforced in this sector. We find that greater import competition from China is associated with an increase in the likelihood of informal enterprises selling their final output to other enterprises directly, or through a contractor. This finding is consistent with formal manufacturing firms outsourcing production activity to informal firms to cut marginal production costs in response to greater import competition. Indeed, we find that the relationship between import competition and outsourcing activity among informal enterprises is mitigated in states with relatively proemployer labour regulation. Lastly, we find that output per worker increases for informal sector firms that outsource, especially in states with pro-worker labour laws. This result expands on the results in McCaig and Pavcnik (2018). They show that a positive trade shock following the United States-Vietnam Bilateral Trade Agreement led to more formalization of the economy through the shrinking of the informal sector. We find that the impact of trade on the informal sector can be heterogeneous - firms that sell to or are on contract to sell to other enterprises gain, but other informal sector firms may not.

Our study makes several contributions in addition to using new and unique data on firm outsourcing. First, we provide strong evidence on trade, especially product market competition, as a determinant of outsourcing activity by firms (McLaren, 2000; Buehler and Burghardt, 2015; Stiebale and Vencappa, 2018). We hence highlight the role of international trade in shaping the organization of firms. Second, our study relates to the literature on the role played by labour market rigidity in spurring firms to outsource production activity in response to trade liberalization (Goldberg and Pavcnik, 2003). This is specially relevant in the case of developing countries that are characterized by large informal sectors, where labour laws are harder to enforce. By increasing the cost of employing workers in a formal setting, rigid labour laws may incentivize firms to outsource activity to the informal sector, particularly in the face of greater foreign competition. By studying the role of labour regulation in this context, we highlight the labour market implications of international trade and the fragmentation of production (Hummels et al., 2014).

 $^{^{22}}$ We thereby address one crucial concern: the endogeneity of labour regulation. Labour laws under the IDA only apply to factories and hence, to jobs related to manufacturing. If our results reflect the effects of labour regulation, we should not see outsourcing of professional jobs responding to the import competition shock. This is indeed what we find.

Finally, our study focuses on south-south trade. Whereas trade theory identifies low-wage countries as a likely source of disruption to high-wage countries' manufacturing firms, Krugman (2008) points out that free trade with countries of any income level may affect the dynamics of the domestic market. A large body of empirical evidence demonstrates that import competition, especially from China, significantly affects the dynamics of manufacturing firms; however, the lion's share of these studies concentrate on developed countries. We investigate the effect of the rise in Chinese imports on outsourcing activities of firms in India, another emerging economy.²³ Ex ante, it is not unreasonable to expect different effects of Chinese import competition on developing countries, given the technological similarity between them and China (di Giovanni et al., 2014).²⁴

The rest of our paper is organized as follows. Section 2 presents the data we use with some stylized facts. Section 3 details our empirical specification and identification strategy. Section 4 presents results studying the relationship between import competition and outsourcing and the role of labour regulation. Section 5 discusses results and mechanisms using a simple analytical framework. Section 6 concludes.

2 Data and Preliminary Analysis

2.1 Firm level Data

Data are drawn from the PROWESS database, constructed by the Centre for Monitoring the Indian Economy (CMIE). The database contains information on approximately 27,400 publicly listed companies, all within the organized sector, of which almost 9000+ are in the manufacturing sector. We use data for around 5,500+ firms, for which there is consolidated data on outsourcing activities. The dataset is classified according to the 5-digit 2008 National Industrial Classification (NIC). We re-classify it to 4-digit NIC 2004 to facilitate matching with other important industry level variables; hence, all categorizations made throughout the paper are based on the 2004 NIC classification. The dataset spans 105 (4-digit 2004 NIC) disaggregated manufacturing industries that belong to 22 (2-digit 2004 NIC) aggregate ones.

The data are captured from annual income statements and balance sheets of all publicly listed companies. Majority of the firms in the data set are either private Indian firms or affiliated to private business groups, whereas a small percentage of firms are either government or foreign-owned.

 $^{^{23}}$ Most studies focus on employment, output, product variety, wages, innovation and productivity as outcomes of interest.

 $^{^{24}}$ di Giovanni et al. (2014), in examining the global welfare impact of China's trade integration and technological change rank ten developing countries in terms of technological similarity to China. Among this group of countries, India is ranked as the country with the closest technological proximity to China; India's technological similarity index being 0.928 to that of China.

The database covers large companies, companies listed on the major stock exchanges and small enterprises. Data for large companies are worked out from balance sheets, while CMIE periodically surveys smaller companies for their data. However, the database does not cover the unorganized sector. The dataset accounts for more than 70% of economic activity in the organized industrial sector, and 75% (95%) of corporate (excise duty) taxes collected by the Indian Government (Goldberg et al., 2010). We use data on all manufacturing firms from 1995 through 2007.

Most importantly, the PROWESS database collects data on outsourcing expenditure incurred by firms. We exploit this unique variable in our empirical analysis. Specifically, we utilize:

(1) information on outsourcing of manufacturing jobs. The dataset reports expenses incurred by firms to get their manufacturing tasks done from outside parties. It includes labour charges, fabrication charges, processing charges, machining charges, fettling charges, conversion charges, contracted production and sub-contracted production. We use this as our main outcome of interest.

(2) information on outsourcing activity of professional jobs. These are expenses incurred by firms for engaging external professional services. The services include: (i) software development fees, (ii) IT enabled service charges, (iii) cost audit fees, (iv) legal charges, (v) miscellaneous professional services, (vi) auditors' fees, and (vii) consultancy fees. We use this as a placebo. Detailed information on variables used in our analysis is presented in **Appendix A**.

In addition to this, the dataset also rolls out information on a vast array of firm level characteristics, including total sales, imports, cost, compensation (wages plus incentives), production factors employed, expenditure, gross value added, assets and other important firm and industry characteristics. Variables are measured in Indian Rupees (INR) million, deflated to 2005 using the industry-specific Wholesale Price Index. CMIE uses an internal product classification that is based on the HS (Harmonized System) and NIC schedules. Around 20% of firms in the data set belong to chemicals, followed by food products and beverages (12.81%), textiles (10.81%) and basic metals (10.46%).

2.2 Stylized Facts: Outsourcing of Manufacturing Activity

In this section, we present stylized facts on outsourcing of manufacturing jobs by Indian firms. **Table 2** shows key firm characteristics by outsourcing status. We compare summary statistics on sales, total assets, gross value added, total factor productivity, export and import volume, R&D intensity for firms involved in outsourcing of manufacturing jobs to firms not involved in outsourcing. Firm involved in outsourcing earn significantly more from sales, are larger, have greater value-added, trade more, adopt better technology (proxied by R&D expenditure), employ

more capital and managerial or skilled workers.

Next, we present results by firms located in states with pro-worker versus pro-employer labour laws. We look at outsourcing, its share in total expenses and the percentage of firms involved, averaged over time (both in the aggregate and by state group) in **Table 3**. These patterns echo that outsourcing activity is more prominent in states with pro-worker labour regulation. We repeat the exercise by type of industry – final goods versus intermediate goods in **Table 4**. The table suggests that outsourcing activity is more prevalent in the case of final good-producing industries relative to intermediate good-producing industries, particularly in states with relatively pro-worker labour regulation.²⁵

Table 5 presents total outsourcing expenditure, share of outsourcing expenditure and percentage of firms involved in outsourcing by industries at the NIC 2-digit level. The table shows substantial heterogeneity in outsourcing activity across industries. Total expenditure on outsourcing in column (1) shows that the expenditure is highest for the automobile industry and lowest for office, accounting and computing machinery. In column (2), we focus on the share of outsourcing expenses in total expenses by a firm; share of outsourcing expenditure is highest in case of labourintensive industries, such as apparel and tobacco products, where it is over 1%, while accounting and computing machinery shows the lowest at 0.02%. Broadly, more labour-intensive industries show a larger share of outsourcing as a share of total expenses. This is consistent with the idea that outsourcing is motivated by lower labour costs outside of formal manufacturing. Lastly, in column (3), the percentage of firms outsourcing ranges from 21 and 20% of firms in fabricated metal products and machinery and equipment to a mere 3% in office, accounting and computing machinery.

Finally, **Table 6** demonstrates the change in distribution of mean outsourcing share in total expenditure in industries between 1992 – 2001 and 2002 – 2007, before and after Chinese accession to the WTO. Relative to 1992 – 2001, a far greater number of Indian manufacturing industries have firms reporting outsourcing shares greater than 0.5% on average in 2002 - 2007, confirming the increase in outsourcing activity in Indian manufacturing post 2002. Overall, our findings in this section support the idea that increased Chinese import competition is associated with greater outsourcing activity in Indian manufacturing firms. We examine this relationship rigorously in our empirical analysis.

²⁵**Table 16** of **Appendix B** shows a more detailed breakdown of outsourcing activity across industries producing basic goods, intermediates, capital goods, consumer durables and non-durables. Outsourcing activity is greatest for consumer durables and non-durables.

3 Empirical Specification

Our goal is to study the impact of increased import competition from China on outsourcing intensity of manufacturing jobs among Indian firms. This section lays out the strategy we use to investigate the effect of China's rising share of exports in the Indian market on outsourcing expenses as a share of total expenses in manufacturing firms. To establish causality between greater Chinese import competition and outsourcing by Indian manufacturing firms, we use China's entry to the WTO on December 2001 as a quasi-natural experiment.

China's accession to the WTO was significantly driven by its movement towards a more marketoriented economy. This transition is a result of the following internal factors: (a) significant ruralto-urban migration of workers, (b) firms/industries gaining access to foreign technologies, capital and intermediate goods that boosted productivity growth and (c) multinational access to operate in the country (Autor et al., 2013). These internal reforms had significant positive effects on China's trade, which eventually led to the country's accession to the WTO.

The economic reforms undertaken by China in the post-1990 period in anticipation of becoming a member of the WTO and integrating into the global economy is an important element of our empirical strategy. Since China's membership to the WTO in 2001 was influenced by factors not related to the activities of Indian firms in their domestic or export markets, it can be interpreted as an exogenous shock from the standpoint of India. Furthermore, there were no trade agreements between India and China in the period prior to accession. It is hence unlikely that Chinese integration into world trade could be confounded with other factors related to the activities of Indian manufacturing firms.

Notwithstanding the assumptions underlying our empirical strategy, there is one important concern that needs to be addressed: the demand for Chinese goods by India, especially after 2001, may have been due to import demand shocks across industries in India²⁶. Failure to address this concern may result in biased coefficient estimates and incorrect inferences drawn from our findings. In order to tackle this issue, we use an empirical strategy similar to Autor et al. (2013) among others.

We estimate the following OLS fixed effects equation as our baseline:

²⁶In case of the US, which we use as a proxy for an alternate export destination, Autor et al. (2013) show that the rise in the Chinese share of imports was not due to import demand shocks in the U.S., but because of an increase in comparative advantage of Chinese goods. Moreover, this increased significantly after 2001.

$$outsourcing_{ijt} = \beta_1 DComp_{IN,jt-1}^{China} + X_{jt-1} + firmcontrols_{ijt-1} + \mu_i + \gamma_t + \theta_j^t + \varepsilon_{ijt}$$
(1)

outsourcing_{ijt} is expenditure on outsourcing of manufacturing jobs as a share of total expenses by firm *i* in sector *j* at time t.²⁷ We define $DComp_{IN,jt-1}^{China}$ as a measure of Chinese competition that an Indian (*IN*) industry (*j*) faces in its domestic market because of the unilateral liberalization policies pursued by China (*China*). To create the $DComp_{IN,jt-1}^{China}$ index, we match the Indian firm level data with HS six-digit product level, destination-specific data for China on import flows. We thus construct a ratio that reflects the amount of competition faced by a firm *i* belonging to industry *j*. We create this index at the NIC 2004 4-digit level using the concordance table by Debroy and Santhanam (1993). It is defined as the share of Chinese imports by India in industry *j* at time *t* divided by total domestic production, imports and exports for industry *j* in 1994. For example, let us consider the Automobile sector (*j*). Then, $DComp_{IN,it-1}^{China}$ can be written as:

$$DComp_{IN,j=Automobile,t-1}^{China} = \frac{M_{IN,j=Automobile,t-1}^{China}}{(Y_{j=Automobile,95} + M_{j=Automobile,95} - X_{j=Automobile,95})}$$
(2)

Therefore, $DComp_{IN,j=Automobile,t-1}^{China}$ is the total amount of Automobile imports from China in a given period, relative to the total production $(Y_{j=Automobile,95})$, total imports $(M_{j=Automobile,95})$ and total exports $(X_{j=Automobile,95})$ of automobiles in the base year 1995. Our hypothesis is that $\beta_1 > 0$, or that greater import competition induces firms to outsource more.

 X_{jt-1} is a set of control variables at the industry level to account for industry specific factors that are related to Chinese import competition and outsourcing intensity jointly. In various specifications, these include the import tariff on the final good produced in sector j, the import tariff on inputs used in sector j (captured by a weighted average of the output tariffs across sectors that supply inputs to j, with input shares as weights), a measure of import competition from China faced by Indian firms in an export destination²⁸, in our case the US²⁹, and share of Indian imports from other low-wage countries.

²⁷Given that our key dependent variable is fractional in nature with a large proportion of zeroes, we also present results from a fractional logit model and PPML to show that our results are robust across these specifications.

²⁸We follow the same method as outlined above in constructing the index of competition that Indian firms face in the US from Chinese imports. We use UN-COMTRADE for data on imports by US industries from the world and China at the 4-digit level. We then match US industries to Indian industries using the International Standard Industrial Classification (ISIC) of all economic activities by the UN.

²⁹ Autor et al. (2013) show that Chinese imports in the US increased significantly after China became a member of

 $firmcontrols_{ijt-1}$ is a vector of variables that includes firm size, age, age squared, and a proxy for the extent of a firm's technology adoption. The extent of technology adoption is measured as the share of R&D expenditure plus royalty payments for technical know-how in gross valueadded (GVA) of a firm. This variable captures technological differences between firms, which can potentially affect outsourcing activity (Acemoglu et al., 2010). We use total sales of a firm as its size indicator. All variables are lagged at (t - 1). μ_i , γ_t are firm and year fixed effects that account for unobserved, firm specific time-invariant and year shocks. θ_j^t are either the interactions between industry fixed effects and year trends or industry-year fixed effects. These account for other potential unobserved factors, such as policy changes or dependency on external finance that may affect outsourcing. We cluster standard errors at the industry level.³⁰

4 Results: Import Competition and Outsourcing

4.1 Baseline

Table 7 presents our baseline results by estimating Eq.(1) using industry-year trends, 2-digit industry-year fixed effects, 3-digit industry-year fixed effects and state-year fixed effects. Columns (1) - (14) use outsourcing expenditure on manufacturing jobs as a share of total expenses as the dependent variable. Column (1) regresses lagged import penetration ratio from China controlling for firm age, age squared, size, technology adoption of a firm and interactions of industry fixed effects at 4-digit level and year trends. Both size and technology adoption are at (t-1) and in real terms. Our coefficient of interest is positive and significant; a 10 percentage point increase in import competition from China increases the outsourcing share of manufacturing jobs in total expenses by 0.13 percentage points. Columns (2) and (3) repeat column (1) but by replacing industry-year trends with industry-year fixed effects at 2-digit and 3-digit levels respectively. These industryyear fixed effects control for other demand shocks, industry specific policies favouring (or not) outsourcing, changes in the pattern of products produced (production of some products involves more outsourcing than others, such as automobiles), contractibility of these industries, thickness of the domestic market for input suppliers, the relative cost of searching (for an outsourcing partner) or of customizing inputs and dependency on external finance.³¹ Our benchmark estimate is robust and positively significant.

the WTO. We also combine US, EU and ASEAN to construct a different version of the export market competition index.

³⁰Note that the observations across different specifications vary as we add control variables.

³¹Boehm and Oberfield (2018) show that contract enforcement is a major factor in understanding how firms source inputs and organize production. Firms in states with weaker enforcement appear to be more vertically integrated.

Columns (4) – (6) include input $(InpTariff_{jt-1})$ and output tariffs $(OutTariff_{jt-1})$ to account for trade liberalization undertaken by India in the 1990s, Chinese import competition faced by Indian firms in a third country (the US) $(FComp_{IN,jt-1}^{China})$, and import competition from other low-wage countries $(DComp_{IN,jt-1}^{Other LWC})$. The impact of Chinese import penetration continues to be robust even after controlling for import tariffs (both input and output tariffs), export market competition, and competition from other low-wage countries. Our estimate remains stable but increases slightly – a 10 percentage point increase in import competition from China now increases the outsourcing share of manufacturing jobs in total expenses by 0.13–0.17 percentage points. We also find limited evidence of export market competition affecting outsourcing positively.

Column (7) additionally introduces an interaction of state and year fixed effects; these interaction terms will help us to control for unobservable state characteristics that may vary over time (like the presence of an informal sector or interlinkages between the formal and informal sectors, state-level laws favouring outsourcing by firms, the contracting environment in each state and financial development) and may influence our outcome of interest. Adding these interaction terms does little to our benchmark result.

Next, we undertake further checks in columns (8) - (14). Column (8) employs a first-differenced specification and finds that the outcome remains the same. Using a different estimation method does little to change our benchmark finding. Another issue that might affect our results is that there is correlation over time in key variables for a given firm. We counter this by running a long difference specification in column (9). We use 1995 as the base year and compare the outcome with 2007. We find a significant positive effect of Chinese import competition in the domestic market on outsourcing activity of Indian manufacturing firms, with no effect for export market competition. In other words, a rise in Chinese import competition in the Indian domestic market significantly induces Indian firms to outsource more manufacturing jobs in 2007 compared to 1995.

Looking solely at Chinese imports by the US as a proxy for export market competition may not reveal the true competitive effect faced by Indian firms in export market(s). To address this possible shortcoming, we construct an index that aggregates the shares of Chinese imports in two other primary export markets for Indian firms, namely the EU and ASEAN, with that of the US. We then substitute the original foreign competition index with the composite index based on these three export market destinations in column (10). In other words,

$$FComp_{IN,jt-1}^{China} = \frac{M_{US,jt-1}^{China} + M_{EU,jt-1}^{China} + M_{ASEAN,jt-1}^{China}}{(M_{US,jt-1}^{World} + M_{EU,jt-1}^{World} + M_{ASEAN,jt-1}^{World})}$$

As the coefficients demonstrate, our benchmark results remain the same – we find strong evidence of outsourcing in response to Chinese competition in the domestic market. We continue to find some weak evidence of competitive effects from export market(s) on outsourcing of Indian firms.

Taking a cue from Burgess and Pande (2005), we use China's joining of the WTO in 2001 as a structural break to estimate a trend break model in column (11) to control for the differential time trends that may affect our outcome variable(s) using the following specification:

$$outsourcing_{ijt} = \beta_1 [DComp_{IN,jt-1}^{China} \times (t - 2001)] + \beta_2 [DComp_{IN,jt-1}^{China} \times (2002 - 2007)] + X_{jt-1} + firmcontrols_{ijt-1} + \mu_i + \gamma_t + \theta_j^t + \varepsilon_{ijt}$$
(3)

Here, (t-2001) is a linear time trend and captures the differential pre- and post-trends of China joining the WTO in 2001, whereas (2002-2007) is a fixed time trend capturing China's membership to the WTO. These terms enter the regression interacted with $DComp_{IN,jt-1}^{China}$. The time trends have a switch in 2001 because of China's membership to the WTO. If China's membership to the WTO in 2001 had significantly influenced outsourcing activity of Indian firms, we expect the interaction term of the [2002 - 2007] trend and $DComp_{IN,jt-1}^{China}$ to be significantly different from the pre-trend interaction.

Our coefficients show that the post-trend is significantly different from the pre-trend. For example, the effect of China's membership to the WTO in 2001 on the share of outsourcing expenditure $DComp_{IN,jt-1}^{China} \times (2002 - 2007)$, is positive and significant as compared to the pre-trend (where it is negative). In other words, the result supports our hypothesis – an increase in China's imports after 2001 significantly affects manufacturing outsourcing of Indian firms.

In column (12), we change our independent variable following Liu and Rosell (2013). Our variable of interest now becomes:

$$DComp_{IN,jt-1}^{China} = \sum_{j} s_{ijt} \frac{M_{IN,jt-1}^{China}}{(Y_{j,95} + M_{j,95} - X_{j,95})}$$

 s_{ijt} is the share of firm *i*'s sales in industry *j* at time *t*. $Y_{j,95}$, $M_{j,95}$, and $X_{j,95}$ are as defined before. Multiplying the import penetration ratio with the sales share of an individual firm transforms the ratio to the firm level. As the estimate of interest demonstrates, changing the independent variable does not induce any change in our finding. We continue to find strong effects of import competition from China.

Since our dependent variable is a ratio, estimating zero-valued variables with OLS may produce biased estimates. Hence, we use a fractional logit model in column (13).³² This method estimates the coefficients in terms of percentage changes and the dependent variable does not need to follow a Poisson distribution or be integer-valued (it can be continuous). As the point estimates demonstrate, the Chinese import penetration ratio continues to significantly increase the share of outsourcing expenses of manufacturing activities in total expenses.

Column (14) restricts the sample to years 1995 – 2001, before Chinese accession to the WTO. We do this as a placebo test, to show that the effect of Chinese import competition on outsourcing comes entirely from the significant increase in Chinese imports that India witnessed after China joined the WTO in 2001. In other words, we should not find any effect of Chinese import competition on the outsourcing share of manufacturing jobs for Indian firms in the 1990s, as the competition did not intensify then. Our conjecture turns out to be true; our coefficient of interest is not significant. Overall, our results indicate that import penetration from China significantly affects the intensive margin of outsourcing.

Even though we use two different estimation methods to control for zeros in our outcome of interest, we also replace our dependent variable with a binary indicator that equals one if the firm reports a positive amount of outsourcing expenses and zero otherwise in **Table 17** (**Appendix B**). We do this for two reasons: (a) to check whether our results hold irrespective of the kind of outsourcing indicator we use, and (b) to check whether import competition from China also affects the extensive margin of outsourcing. Also, such a binary variable might be less vulnerable to measurement error relative to our main variable.

The change of dependent variable does not alter our benchmark finding. Columns (1) - (12) present results from the extensive margin of outsourcing – where the dependent variable takes a value 1 if the outsourcing expenditure on manufacturing jobs of a firm is greater than zero. Our coefficient of interest continues to be positive and statistically significant across all specifications. Put together, our results show a strong positive relationship between Chinese import competition and outsourcing of manufacturing activity by Indian manufacturing firms.

Before proceeding to check for further sensitivity of our benchmark finding, we investigate

 $^{^{32}}$ We also use a Poisson Pseudo-Maximum Likelihood (PPML) model following Silva and Tenreyro (2006). Results remain qualitatively similar.

whether the effect of Chinese import competition is a spillover effect from the trade reforms of the 1990s undertaken by India in response to a balance-of-payments shock. Results are presented in **Table 18** (**Appendix B**) – we regress input and output tariffs on the share of outsourcing expenditure. We do not find any robust effect of either the input or output tariff on outsourcing expenditure by Indian manufacturing firms.

IV Analysis While in principle it is useful to use a lagged independent variable as a proxy for the contemporaneous import penetration index to tackle the simultaneity problem, it could still be endogenous. For example, an increase in the demand for particular products in India after 2001 may trigger a disproportionate increase in Chinese imports in these product categories and simultaneously impact Indian firms producing them. This could also be true for unobserved technology shocks common to both countries, like innovation in labour cost-saving technology (Utar and Torres-Ruiz, 2013). Such endogeneity can bias the estimate of the impact of Chinese import competition on outsourcing.

To overcome the possible endogeneity concern(s), we follow Autor et al. (2013), Acemoglu et al. (2016) in instrumenting for Chinese imports to India by Chinese imports to other similar developing countries. The instrument for (2) is computed as:

$$DComp_{BIMM,jt-1}^{China} = \frac{M_{Others,jt-1}^{China}}{(Y_{j,95} + M_{j,95} - X_{j,95})}$$
(4)

where $M_{Others,jt-1}^{China}$ is the lagged value of Chinese imports by an industry in Brazil, Indonesia, Malaysia and Mexico. This approach assumes that the rise in Chinese manufacturing exports to other developing countries was primarily driven by internal supply shocks and reduced trade costs, but not by unobserved import demand shocks in developing countries (Autor et al., 2013). The Chinese share of imports by Brazil, Indonesia, Malaysia and Mexico must be exogenous from the perspective of Indian firms as it is expected to be driven by China. In other words, Chinese exports to these countries are likely to be correlated with Chinese exports to India but not with Indian conditions driving Indian imports.

Results from the IV estimations along with their first-stages are presented in **Table 8**. Our IV results qualitatively mirror results in **Table 7**, though the magnitudes of the coefficient of interest across columns are larger. This is possible if unobserved factors driving outsourcing activity by Indian firms and imports from China simultaneously lead to inconsistent estimates of the impact of Chinese import penetration on outsourcing. Columns (1) through (4) present results for outsourcing

as a share of total expenses with column (5) presenting results for the time period 1995-2001. Overall, our IV results suggest that a 10 percentage point increase in the Chinese import penetration ratio increases the share of outsourcing in total expenses by 0.24-0.50 percentage points. We continue to find no effect of Chinese import penetration in the pre-2001 period.

Having now established that our results hold irrespective of the type of import competition index, the method or specification we use, we proceed to account for other potential endogeneity concerns in our estimation. In the next section, we control for several other import competition indices and for access to intermediate inputs.

Competitive Pressures from Other Regions and The Case of Intermediate Inputs Our

result that import competition from China increases outsourcing by Indian firms may be due to overall import competition, including from other destinations. In order to delve into this, we calculate a general import competition index (World), and indices for high-income countries (High-Income), North America (NA), European Union (EU), Latin American countries (LA), least-developed countries (LDC), Middle-east and North African countries (MENA), and South Asian countries (SA). Results using these indices as measures of import competition are presented in columns (1) – (5) of **Table 9**. We start by using a general import competition index – $DComp_{IN,jt-1}^{World}$ along with $DComp_{IN,jt-1}^{China}$ in column (1). The coefficient on Chinese import penetration is statistically significant and positive, suggesting that it is not import competition per se, but import competition from China that is associated with more outsourcing of manufacturing jobs by Indian firms.

Across columns (2) through (5), we show that this positive and significant relationship is robust to controlling for import competition from High - Income (column (2)), NA and EU (column (3)), LA, LDC, MENA, and SA (column (4)) countries and all of these put together (column (5)). We find that the positive relationship between Chinese import competition and outsourcing of manufacturing jobs endures.

Another factor that might affect our findings significantly is the way we use total imports in our estimations; in other words, it includes imports of intermediate inputs by Indian firms (Iacovone et al., 2013). For example, imported intermediate inputs from China may be cheaper and of higher quality than locally sourced inputs, lowering production costs of the firm and allowing it to outsource more. To account for this possibility, we generate a measure of the share of imported inputs from China by Indian firms using Indian input-output (I-O) tables in columns (6) - (8).³³ We weight the I–O coefficient of each sector (at NIC 4-digit level) that is used as an input by its import share,

 $^{^{33}}$ We use the 1999 I-O table to choose input coefficients for each of the 2004 NIC 4-digit sectors. We additionally test for robustness by substituting with the 1993 I-O table and find that the results remain.

and then by the Chinese share in imports for that sector. By summing these measures, we arrive at a measure, *InpDComp*, that gives us the average weighted sum of intermediate goods imported from China at a sectoral level, where the weights are given by the coefficients of the I-O table.

If Chinese import competition in upstream industries is correlated with import penetration in the final goods sector, then our coefficient of interest might be inconsistently estimated. Estimates from columns (6) – (8) show that our main result remains robust to the addition of this control variable. We do not find any effect of imported intermediate goods, InpDComp, from China. It is product market competition that induces firms to outsource a larger share of their manufacturing activities.

Heterogeneous effects across industries and firms Table 10 examines if the increase in outsourcing expenditure of a firm in response to greater import competition is differential across industries based on various industry characteristics. For all our specifications in Table 10, we control for state-year and industry-year fixed effects at the 3-digit level.

Column (1) asks if the impact of import competition on outsourcing varies across industries based on skill-intensity. There is considerable debate in recent times on whether international trade has contributed to the declining fortunes of less skilled workers. Feenstra and Hanson (1996) argue that outsourcing (by which they mean import of intermediate inputs) has led to an increase in relative demand for skilled labour in the US because firms may respond to import competition from low-wage countries (such as China) by restructuring production towards skill-intensive activities. This may apply particularly to industries intensive in skills. Our idea in Column (1) is to understand whether this applies to the Indian case. We introduce skill-intensity (defined as the ratio of the number of non-production workers to total employees of an industry) in the pre-2001 period and its interaction with Chinese import penetration as additional variables in the baseline specification. We do not find any evidence that industries that were initially intensive in highly skilled labour are engaged in more or less outsourcing.

Column (2) uses factories (at the initial level) and its interaction with Chinese import competition as additional variables. The idea is to examine if industries with fewer firms respond differently to import competition relative to industries with a large number of firms. We find no evidence that this is the case. Column (3) checks whether highly productive firms outsource more in response to greater import competition. We calculate total factor productivity of a firm using the Levinshon-Petrin (2003) methodology and use the initial level of productivity and its interaction with Chinese import penetration as additional variables. Our estimate shows our conjecture to be true – initially productive firms outsource more as a result of Chinese competition in the domestic market. This is consistent with Grossman and Helpman (2004).

Firms producing mutiple products as opposed to a single product may outsource more in response to import competition, as they rationalize their products. We classify firms according to the number of products they produce and divide them into two categories – single- and multi-product firms in column (4). We create a dummy variable MPFirm which takes a value 1 if a firm is a multi-product firm ³⁴. We interact MPFirm with $DComp_{IN,jt-1}^{China}$ and show that the impact of import competition is magnified for multi-product firms.³⁵

Next, Grossman and Helpman (2004) argue that managerial incentives may be positively correlated with outsourcing. Managers who oversee outsourcing of production and assembly activities are offered high-powered incentives in order to facilitate outsourcing in an efficient manner. We find strong evidence for this in Columns (6) – (10).

Column (6) uses total managerial compensation (compensation is defined as wages plus incentives) of a firm, whereas column (7) uses total managerial wages interacted with Chinese import penetration as additional variables. We do not find any evidence that firms paying either higher compensation or wages react differently to import competition. Columns (8) – (10) use interactions with total managerial incentives and incentives divided into top management (executives) and middle management levels (non-executives or directors). The latter two comprise the managers' group. Executives (directors) are defined as managers with (without) executive powers. Executives include, for instance, the CEO, CFO and Chairman, whereas Directors may cover positions such as Divisional Managers. We find that with greater import competition from China, outsourcing increases by more in firms paying higher managerial incentives.

In **Table 19** (**Appendix B**), we explore heterogeneous effects of import competition on outsourcing across firm types. We interact our main Chinese import penetration variable with indicator variables for size categories in Column (1), whether the firm is in a final good or intermediate good industry in column (2), if the firm is an exporter or not in column (3) and whether the firm is a foreign or domestic firm in column (4).

We find strong evidence of an impact of import competition on outsourcing across the size distribution of firms. However, the effect is about 30% higher for large firms. In addition, we find that the impact of import competition on outsourcing is concentrated among firms in final

³⁴Though the Prowess database contains information on products produced by firms, outsourcing expenditure is not available at the product level. Hence, we are unable to conduct our analysis at the firm-product level, or to ascertain if firms outsource their core or peripheral products.

³⁵In an extended analysis, we find that a significant, positive relationship exists between import competition and outsourcing for single-product firms that are exporters and that produce final goods.

good industries, firms that are non-exporters and domestic firms. This is likely to be the case if firms that are oriented internationally have to conform to international norms and standards in their technique of production (capital-labour ratios), have to demonstrate adherence to labour standards or are subject to more labour inspections from state officials (Sundaram et al., 2017).

Other Extensions Our analysis focuses primarily on the 1-year lagged, short-run effects of import competition from China. This suggests that the observed impact may be an outcome of changes that occur within relatively short timeframes. Nonetheless, import competition may also affect the share of outsourcing via changes that are expected to occur over longer timeframes, such as general equilibrium adjustments of prices, outputs, or even opportunities for more outsourcing. To examine the role of dynamics, we estimate the following model:

$$outsourcing_{ijt} = \beta_1 DComp_{IN,it-n}^{China} + X_{jt-1} + firmcontrols_{ijt-1} + \mu_i + \gamma_t + \theta_i^t + \varepsilon_{ijt}$$
(5)

where $n \in [0, 3]$. This specification is equivalent to Eq.(1), but it considers the impact of Chinese import competition over different periods, ranging from its contemporaneous effect (n = 0), to its impact in t - 3 (n = 3). Our focus is on β_1 . Yet, given that the sample is more restricted under these specifications, we place greater emphasis on interpreting magnitudes, rather than precision. Results appear in **Table 20** (**Appendix B**). Columns (1) - (4) test the cases of 0, 2, and 3-year lags for the share of outsourcing expenditure as the dependent variable, whereas columns (5) - (8)do the same for the extensive margin of outsourcing. In all cases, the sign of β_1 is similar to the one estimated in the baseline specification. However, the estimated magnitudes suggest that any potential contemporaneous effects appear to be equivalent to the ones observed in the baseline; conversely, the effects of an increase in import penetration from China in 2- and 3-year lags appear to be greater than those we observe in the baseline. This suggests that both short and long term changes are applicable to some extent, yet the former is more dominant, given that the effect of the 1-year lag appears stronger and more robust.

4.2 Role of Labour Market Regulation

India is a federal democracy and under the Indian Constitution of 1949, industrial relations is a concurrent subject. This implies that central and state governments have joint jurisdiction over labour legislation. The key piece of central legislation is the IDA 1947, which sets out the conciliation, arbitration and adjudication procedures to be followed in the case of an industrial dispute. The Act was designed to offer workers in the organized sector some protection against exploitation by employers (for details, see Besley and Burgess, 2004).³⁶ It has been extensively amended by state governments during the post-Independence period. Besley and Burgess (2004) code all 113 such amendments since the Act was passed and designate them as being either "neutral", "proworker", or "pro-employer" to investigate how labour regulation impacts economic performance at the state-level.³⁷

The most controversial laws deal with the conditions for hiring and retrenching of workers and with the closure of establishments. For example, a 1976 amendment to the IDA 1947 made layoff, retrenchment and closure illegal except with the previous permission of the appropriated government for all firms with more than 300 workers. This coverage was subsequently extended in 1982 to all firms with more than 100 employees.³⁸

We exploit this variation across Indian states to ask if import competition impacts outsourcing differentially for firms located in states with pro-worker, as opposed to pro-employer labour regulation, with neutral states coded as pro-worker. We posit that restrictions on hiring and retrenchment of workers, shift work and closing down of factories act as an implicit tax on employing labour in-house in the formal sector. A large literature has emphasized the role played by rigid labour markets and stringent labour market regulation in pushing up implicit labour costs in developing countries (Besley and Burgess, 2004), particularly in the formal sector, where labour laws are enforced.³⁹ In a couple of recent studies, Adhvaryu et al. (2013) and Chaurey (2015) use the same classification to investigate the effect of demand shocks on total industrial employment and employment of contract labour, respectively, and find that in response to demand shocks, firms in states with pro-worker labour regulation react differently.

However, before proceeding to the estimations, we test for one crucial identifying assumption – we compare firm outsourcing across these two states before China joined the WTO in 2001 and show that there were no differential time trends. **Figure 3** plots the normalized share of expenditure on outsourcing of manufacturing jobs in total expenses of a firm for both states with pro-employer and

³⁶The Act is comprised of seven chapters and forty sections, specifying the powers of government, courts and tribunals, unions and workers and the exact procedures that have to be followed in resolving industrial disputes.

³⁷Although all states have the same starting point, they diverge from one another over time.

³⁸In addition, some states further amended Chapter Vb above and beyond what is specified in the central Act. For instance in 1980, West Bengal extended Chapter Vb to firms hiring 50 or more workers.

³⁹One strand of literature has found negative economic impacts of amending the IDA regulations that make it harder to fire workers—lower output, employment, investment, and productivity in formal manufacturing (Besley and Burgess, 2004; Aghion et al., 2008; Ahsan and Pages, 2009). On other hand, other scholars have questioned whether amendments made to the IDA have indeed increased flexibility in firing (Bhattacharjea, 2006) or whether these regulations have even been enforced (Nagaraj, 2002).

pro-worker labour laws. The plot shows that there is no clear differential pattern of outsourcing between these states before 2001 – the difference starts to grow along with the increase in import competition from China. Firms located in states with pro-worker labour laws start to outsource more than firms in states with pro-employer labour laws after 2001.

Using the classification by Gupta et al. (2009) and/or Adhvaryu et al. (2013), we test whether firms in pro-worker labour regimes outsource more in response to Chinese import competition. We estimate:

$$outsourcing_{ijt} = \beta_1 DComp_{IN,jt-1}^{China} + \beta_2 (DComp_{IN,jt-1}^{China} * LMktR_s) + X_{jt-1} + firmcontrols_{ijt-1} + \mu_i + \gamma_t + \theta_j^t + \varepsilon_{ijt}$$
(6)

 $LMktR_s$ is a dummy variable that equals one if labour laws in a state in which firms' are registered are pro-employer. $LMktR_s = 1$, when s = Andhra Pradesh, Karnataka, Rajasthan, Tamil Nadu, and Uttar Pradesh.⁴⁰ On the other hand, $LMktR_s = 0$, when s = Gujarat, Maharastra, Orissa, and West Bengal and for netural states Assam, Bihar, Haryana, Jammu and Kashmir, Punjab, Kerala and Madhya Pradesh. All other variables remain the same as in equation (1), except for X_{jt-1} . We also include interaction terms of all other controls with $LMktR_s$. Table 21 in Appendix B lists the names of states according to their labour regime.

Our main variable of interest now is β_2 – the coefficient on the interaction between $LMktR_s$ and $DComp_{IN,jt-1}^{China}$. It captures the differential effect of Chinese import competition on firms in states with more pro-employer labour laws relative to other states. A positive β_2 would imply that an increase in Chinese import competition induces firms located in states with pro-employer labour laws to increase their outsourcing expenditure more than firms located in other states; vice-versa for $\beta_2 < 0$. We expect β_2 to be negative. In other words, if costs imposed by labour regulation(s) spur firms to outsource manufacturing activity, we expect the interaction term between Chinese import penetration and the indicator for states with pro-employer labour regulation to be negative.

Results are reported in **Table 11**. Overall, we find that compared to firms in pro-employer labour regimes, those in restrictive labour regimes engage in more outsourcing in response to

⁴⁰This is the classification by Gupta et al. (2009). We also check our results using the classification by Adhvaryu et al. (2013), where the "pro-employer" states are – Andhra Pradesh, Karnataka, Kerela, Madhya Pradesh, Rajasthan, and Tamil Nadu.

Chinese import penetration. Column (1) regresses the share of outsourcing of manufacturing jobs on $DComp_{IN,jt-1}^{China}$ and its interaction with $LMktR_s$ controlling for industry-year trends at the 4-digit level. Our results show that a 10 percentage point increase in the share of import penetration from China increases outsourcing share by 0.20 percentage points. Importantly, this is 0.17 percentage points lower for firms in states with pro-employer labour laws. Columns (2) and (3) additionally introduce interactions between industry-year fixed effects at 3-digit level and state-year fixed effects to control for unobservables at the industry and state level, respectively. Using these additional fixed effects does not alter our finding – labour regulation acts as an important channel in determining the relationship between trade and outsourcing. Firms operating in states with pro-worker labour laws outsource more, potentially in order to circumvent them.

We introduce interactions of the labour regulation indicator and all other controls capturing the effects of various dimensions of trade – input tariffs $(InpTariff_{jt-1})$, output tariffs $(OutTariff_{jt-1})$, export market competition $(FComp_{IN,jt-1}^{China})$, and import competition from other low-wage countries $(DComp_{IN,jt-1}^{Other LWC})$ in column (4). β_2 continues to be negative and significant, while β_1 is positive and significant. Columns (5) and (6) divide the sample of firms into single- and multi-product firms, respectively. The coefficients show that the aggregate effect is driven by multi-product firms operating in states with pro-worker labour regulation.

One concern with the interpretation of our coefficients could be that labour regulation is correlated with other factors that determine how firms respond to greater import competition. For example, if workers lobby for pro-worker regulations, states with more manufacturing (or a large blue-collar lobby) may have enacted more pro-worker legislation. Or, firm responses to import penetration shocks may vary by their capital intensity, and labour laws may be correlated with the average capital intensity of firms.

Jayachandran (2006) and Adhvaryu et al. (2013) address such concerns by including relevant area characteristics and their interactions with the main variable of interest. We follow a similar strategy and control for the interaction of baseline characteristics of states with $DComp_{IN,jt-1}^{China}$, including the ratio of production workers, per capita salary of production workers, per capita NSDP (Net State Domestic Product), total tax revenue, total grants received by the state government from the federal government, total expenditure, total expenditure on development and headcount ratios. Column (7) presents our results, which continue to be robust to the inclusion of state level characteristics. In fact, our coefficient estimates increase significantly.⁴¹

⁴¹**Table 22** (**Appendix B**) also checks for the role of labour market regulation in determining the impact of import competition on the extensive margin of outsourcing – the dependent variable takes a value 1 if the outsourcing expenditure of a firm on manufacturing jobs is greater than zero. Our benchmark result continues to hold – firms

Lastly, we test for the robustness of our main finding by using the classification and following the empirical strategy of Adhvaryu et al. (2013) and/or Chaurey (2015):

$$outsourcing_{ijt} = \beta_1 DComp_{IN,jt-1}^{China} + \beta_2 (DComp_{IN,jt-1}^{China} * pro - wor \ker) + \beta_3 (DComp_{IN,jt-1}^{China} * neutral) + X_{jt-1} + firmcontrols_{ijt-1} + \mu_i + \gamma_t + \theta_j^t + \varepsilon_{ijt}$$

$$(7)$$

In this case, pro - wor ker states are Gujarat, Maharastra, Orissa, and West Bengal. The *neutral* states are Assam, Bihar, Haryana, Jammu and Kashmir, Punjab and Uttar Pradesh. The pro - employer states are Andhra Pradesh, Karnataka, Kerala, Madhya Pradesh, Rajasthan and Tamil Nadu as the omitted category. Thus, β_2 and β_3 measure the effect of Chinese import penetration in pro - wor ker and *neutral* states, respectively, relative to pro - employer states. Our primary coefficient of interest is β_2 . We expect β_2 to be positive and significant. For example, suppose that the average effect of Chinese imports is positive, or $\beta_1 > 0$, then a positve estimate of β_2 would imply that relative to pro - employer states. For β_3 , it could be positive, but should be less than β_2 . Column (8) estimates the above equation. We find our hypothesis to be true – the increase in aggregate outsourcing is driven by states with pro-worker labour laws and the increase in outsourcing in "neutral" states is greater than in pro-employer states, but less than in pro-worker states. A 10 percentage point change in the import penetration ratio increases outsourcing by 0.46 percentage points more in states with pro-worker labour regulations relative to states with pro-employer labour regulations.

Placebo We take one further step to ensure that our results are not contaminated by the endogeneity of labour regulations. We use the fact that labour laws under the IDA only apply in case of manufacturing jobs and do not apply to professional workers who perform skilled tasks. If the mechanism we have in mind explains the differential relationship between import competition and outsourcing in states with pro-worker labour regulation, we would not expect to find it for outsourcing of professional jobs. We estimate a placebo regression following equation (6), but by replacing our outcome variable of interest by outsourcing of professional jobs.

located in states with pro-worker labour laws engage in more outsourcing than firms in states with pro-employer labour laws in response to import competition.

Results are reported in **Table 12**. Columns (1) - (5) focus on the share of expenses on outsourcing of professional jobs in total expenses and column (6) uses a dummy variable indicating outsourcing (it takes a value 1 if the outsourcing expenditure on professional services is greater than zero). All coefficients show that there is no differential relationship between Chinese import competition and outsourcing of professional jobs in states with relatively pro-worker labour regulation. Combined, our results provide support for the idea that greater import competition is associated with greater outsourcing of manufacturing jobs, particularly under stringer labour regimes that drive up the relative cost of operating in the formal sector in developing countries.

4.3 Informal Sector

A common feature of developing countries in Africa, Latin America and South Asia, is the presence of a large informal sector.⁴² According to an estimate by Charmes (2012), the informal sector in India contributed around 46% of non-agricultural GVA and 38% of total GDP in India. In terms of employment, according to a recent report by ILO (2018), close to 81% of all employed persons in India make a living by working in the informal sector. Ulyssea (2018) points out that the presence of an informal sector has two contrasting implications: on the one hand, it can lead to widespread evasion of taxes, misallocation of resources and TFP and on the other, it can be beneficial to growth as it can provide flexibility for firms that may be constrained by strict regulations. Our previous results show that in response to import competition, formal sector firms outsource more, especially in states with pro-worker labour laws. Therefore, understanding whether import competition effects on the informal sector support this story is one of the central questions of our paper.

To do so, we exploit a dataset that contains detailed information on informal (unorganized) sector manufacturing enterprises (micro) from the National Sample Survey Organization (NSSO), India. Our data comprises of two rounds of a nationally representative survey of informal enterprises that employ fewer than ten workers for the years 1999-00 and 2004-05.⁴³ The survey asks these enterprises two relevant questions that we utilize for our purpose. First, if enterprises are primarily on contract to sell their product to another enterprise or to a middleman/contractor. Second, if the destination of their final product is another enterprise (as opposed to the consumer). Note that formal sector firms in our data are most likely to outsource manufacturing tasks to firms in the informal sector if their primary motivation is to reduce marginal cost when faced with competition

 $^{^{42}}$ In Brazil, nearly two-thirds of businesses, 40% of GDP, and 35% of employees, in Colombia 50% of workers and 41.9% of GDP and in Mexico 60% of workers and 31.9% of GDP are informal (Ulyssea, 2018).

⁴³This dataset is available every five years. We do not include the 2009-10 round in our analysis since it would also capture the impacts of the financial crisis.

from China. If this is true, we should see a corresponding increase in the likelihood of informal sector firms writing/engaging in a contract or selling their output to other firms in response to greater Chinese import competition.

Using answers from these survey questions, we construct three alternate indicators of outsourcing activity for informal sector firms that take on a value of 1 if (a) a micro-enterprise in the informal sector is on contract to sell a large proportion of output to another firm or a middleman/contractor; (b) the enterprise reports selling most of its output to other enterprises (as opposed to the government or private households); and (c) a combination of the first two, such that the indicator equals one if either the first or the second indicator equals one. We use the last as our preferred indicator.

Table 13 presents our results. We compare the likelihood of an informal sector enterprise either entering into a contract with another enterprise or selling a large proportion of its output to them between the years 1999-00 and 2004-05. Columns (1) - (6) use an indicator which takes a value one if an informal enterprise answers 'Yes' to either question. All regressions include interactions of industry fixed effects and year trends and state-year fixed effects. Overall, our results show a strong, statistically significant and positive relationship between Chinese import competition and the likelihood of engaging in outsourcing.

Column (1) runs estimates a linear probability model. A 10 percentage point increase in Chinese import penetration leads to an increase of 44 percentage points in the likelihood of outsourcing. Columns (2) and (3) check for robustness by using probit and logit methods. The coefficient of interest remains qualitatively the same. We include an interaction between the import penetration ratio and labour market regulation, $DComp_{IN,jt-1}^{China} * LMktR_s$, in columns (4) – (6). We find similar effects as before – informal sector enterprises located in states with pro-worker labour regulation have higher likelihoods of outsourcing. This effect is significantly higher for informal sector enterprises located in rural rather than urban areas – by about 3.5 times. Results are consistent with our hypothesis that costs imposed by stringent labour regulation induce formal sector firms to outsource manufacturing tasks to the informal sector. Columns (7) and (8) divide the composite indicator and show that the result is robust to using alternate indicators.

One important implication of these results is that higher outsourcing to informal sector enterprises might increase the size of the sector and otherwise impact its performance. If it does so, we could then argue that outsourcing is a potential channel through which greater import competition leads to gains across different sectors of the economy, especially a developing one. To understand whether such is the case, we use output per worker as the outcome of interest in columns (9) and (10).⁴⁴ We find that Chinese import competition significantly increases output per worker of informal firms that are engaged in outsourcing, particularly in states with pro-worker labour market regulation. Our results complement the work by McCaig and Pavcnik (2018), who show that an export market shock can lead to a reallocation of workers from the informal to formal sector leading to a contraction of the informal sector.⁴⁵ Our results suggest that the impact of trade on informality can be heterogenous accross informal sector firms, with those that engage in sub-contracting experiencing a boost relative to other firms.

5 Discussion of Results

To summarize, we find that an increase in import competition, particularly, a higher degree of import penetration from China increases a firm's share of expenses on outsourcing of manufacturing jobs in total expenses. We see a similar increase in the extensive margin of firms that outsource. This result is persistent, economically meaningful, robust to a myriad tests, and relatively dominant in the short run. Digging deeper, the analysis points out that labour regulation plays a role in mediating the relationship between trade and outsourcing. Firms that operate in pro-worker labour regimes drive the patterns observed. All these findings are driven by multi-product firms. Last, import competition increases the likelihood of an informal sector micro-enterprise signing a contract to sell to another enterprise or middleman.

We now present a conceptual framework that suggests one mechanism whereby higher import competition leads to greater outsourcing by manufacturing firms, particularly when employing in-house labour is costly. We then explore the implications of this framework empirically.

Analytical framework In this section, we provide a conceptual framework to examine the impact of import competition on outsourcing following Lommerud et al. (2009). Consider a firm i operating in a monopolistically competitive environment producing a variety of a differentiated good, which it produces by using a continuum of inputs indexed by $j \in [0, 1]$. One unit of the final good requires γ_i^{-1} units of each input for firm i. Each input can either be produced in-house or outsourced. In-house, the firm can produce one unit of j using one unit of labour at an exogenous wage rate w. Alternatively, the firm can outsource production at the cost of c per unit of input, where we assume w > c to capture the idea that the wage rate is higher than the marginal cost of outsourcing to smaller (informal sector) firms. For instance, since it operates in the formal sector,

⁴⁴We use total employment as an alternative outcome of interest - the result remains the same.

⁴⁵Ulyssea (2010) also finds that a decline in entry cost in the formal sector reduces the size of the informal sector and improves overall labour market performance.

the firm has to ensure adherence to safety standards, offer benefits, including overtime and abide by hiring and firing regulations. As pointed out by Besley and Burgess (2004), these provisions of the Indian Factories Act do not apply to firms hiring fewer than 10 workers operating in the unorganized or unregistered (informal) manufacturing sector.

An important ingredient of our framework is that outsourcing incurs fixed costs, which depend on the input j. Specifically, ordering the inputs on [0, 1] so that g(j) < g(l) for j < l, the cost of outsourcing k inputs is given by

$$G(k) = \int_0^k g(j)dj \tag{8}$$

Assume that G'(k) > 0 and G''(k) > 0, G'(0) = 0 and $G'(1) \to \infty$, where the last assumption means that it is not economical to outsource all production. A motivation for outsourcing costs increasing exponentially is the co-ordination costs involved in dealing with multiple small firms or contractors.

Demand for the final good is given by $y_i = \Gamma p_i^{-\sigma}$, where p_i is the price of variety *i* and $\Gamma > 0, \sigma > 1$. Suppose that the firm outsources the production of k_i inputs, its profits are given by

$$\pi_i = [(p_i - \gamma_i^{-1}(k_i c + (1 - k_i)w)]y_i - G(k_i)$$
(9)

Substituting for output, we get

$$\pi_i = [(p_i - \gamma_i^{-1}(k_i c + (1 - k_i)w)]\Gamma p_i^{-\sigma} - G(k_i)$$
(10)

The first order condition with respect to price is given by

$$\frac{\delta \pi_i}{\delta p_i} = \Gamma[(1-\sigma)p_i^{-\sigma} + \sigma \gamma_i^{-1}(k_i c + (1-k_i)w)p_i^{-\sigma-1}] = 0$$
(11)

$$p_{i}^{*} = \frac{\sigma}{\sigma - 1} \gamma_{i}^{-1} (k_{i}c + (1 - k_{i})w)$$
(12)

The first order condition with respect to outsourcing at optimal p_i^* is given by

$$\frac{\delta \pi_i}{\delta k_i} = -\Gamma p_i^{*-\sigma} \gamma_i^{-1}(c-w) - G'(k_i) = 0$$
(13)

and the second order condition at the optimal outsourcing intensity k_i^* by

$$\frac{\delta^2 \pi_i}{\delta k_i^2} = \sigma \Gamma p_i^{*-\sigma-1} \gamma_i^{-1} (c-w) \frac{\delta p_i^*}{\delta k_i^*} - G''(k_i^*) < 0$$

$$\tag{14}$$

Given this framework, we can present the following proposition:

Proposition 1
$$\frac{\delta k_i^*}{\delta \sigma} > 0$$
 as long as firm productivity γ_i is above a certain threshold γ_i^* .

A proof of this proposition is presented in **Appendix C**. The intuition here is that an increase in the elasticity of demand is associated with a decrease in price charged and an increase in quantity produced by firms. This increase in quantity makes the lower marginal cost from outsourcing more attractive, inducing the firm to outsource more by incurring the higher fixed costs of outsourcing.

In **Figure 4A**, we provide a numerical example of the relationship between increasing σ and optimal outsourcing intensity for firms with low, medium and high productivity respectively ($\gamma_i = 3, 12 \text{ and } 15$). We set w=5, c=4 and Γ normalized to one. **Figure 4B** shows a similar graph for the relationship between σ and optimal price p_i^* and shows how the price charged declines with greater import competition.

We now consider the differential relationship between import competition and outsourcing in states with pro-worker and pro-employer labour regulation. Figure 5A plots the relationship between σ and outsourcing intensity when w=4.5 and 5.5 respectively. The idea is that in states with pro-worker labour regulation, the cost of hiring labour in the formal sector is higher than in more pro-employer labour regimes. The graph shows that an increase in σ is associated with a much larger increase in outsourcing when the formal wage is higher at 5.5 (equivalently, when the firm is in a pro-worker state). Figure 5B presents a similar figure for the differential relationship between trade and prices for firms in states with pro-worker versus pro-employer labour regimes. While the price decline is initially sharper in states with pro-employer labour laws, for higher values of σ , the decline is sharper in states with pro-worker labour laws. Next, we use firm-product level data to examine this proposition empirically.

Empirical Testing We now compare the impact of Chinese import competition on costs, prices and total sales of firms engaged in outsourcing and not, differentially in states with pro-worker versus pro-employer labour laws. Our hypothesis is that import competition induces firms to lower prices and expand output, thereby incentivizing them to incur the fixed costs of outsourcing production to avail of lower marginal costs; particularly in states with a pro-worker labour regime. Hence, for firms that are engaged in outsourcing, greater import competition should be associated with a decrease in costs and price and an increase in firm size (sales). Also, if the driving factor behind cost gains is labour regulation, we should see these impacts magnified in states with more stringent labour regulation (attentuated in states with pro-employer labour regulation). We present the results in **Table 14**. Columns (1) - (4) use marginal costs, whereas columns (5) - (8) present results for the price charged by firms, and columns (9) - (12) focus on firm sales as the dependent variable, respectively for firms in states with pro-worker and pro-employer labour regimes that outsource and do not outsource.⁴⁶ Results are remarkably in line with our conceptual framework. Chinese import penetration is negatively associated with firm costs for firms that outsource and this effect is magnified for firms in states with pro-worker labour regulation; the same applies for the prices that firms charge. In addition, our results show that greater import competition is associated with larger firm sizes for outsourcing firms, particularly in states with pro-worker labour regulation.

The findings of this study can shed light on central issues related to firm level impacts of import competition; specifically, those related to effects on a firm's boundary, strategy, efficiency, productivity, and growth. This is suggested by a growing body of literature that links vertical integration and outsourcing to firm productivity, performance, and growth. Some studies that provide evidence for this include Grossman and Helpman (2002), Antras and Helpman (2004), Chen et al. (2004), Acemoglu et al. (2010), Hart and Holmstrom (2010), Chongvilaivan and Hur (2012), Legros and Newman (2013), Alfaro et al. (2016), and Boehm and Oberfield (2018). For instance, Alfaro et al. (2016) examine the impact of tariffs on firm boundaries. The empirical results provide strong support for the view that higher output prices generate more vertical integration. Boehm and Oberfield (2018) document that in industries that tend to rely more heavily on relationship-specific intermediate inputs, plants in states with more congested courts shift their expenditures away from intermediate inputs and appear to be more vertically integrated. While our paper looks at similar issues, we probe a new channel that is relatively understudied – import competition and its interaction with labour regulation.

6 Conclusion

Understanding the effects of globalization on a firm's boundary is of first-order importance. Previous research indicates that trade can induce firms to vertical intergrate. However, the literature overlooks the effects of import competition and how labour market regulation can play a role in determining how firm outsourcing responds to trade shocks. This may be prominent in light of the emerging literature on the link between trade liberalization and firm organization, and its effects on productivity and growth, especially in developing economies. This paper attempts to fill this gap.

⁴⁶We follow de Loecker et al. (2016) to estimate marginal costs and prices.

Adopting the case of India, we ask if import competition affects outsourcing. We explore the differential effect of trade on outsourcing across firms located in pro-worker versus pro-employer labour regimes. Using a rich firm level dataset that uniquely reports expenses incurred by firms on outsourcing of manufacturing activities in the Indian manufacturing sector, and exploiting China's accession to the WTO in 2001 as a quasi-natural experiment, we establish a causal link between import competition and the share of outsourcing expenses in total expenses. In addition, Indian labour laws vary significantly across states. We exploit this variation to establish that import competition is associated with greater outsourcing in states with pro-worker labour regulation that potentially increases the cost of employing labour in-house in the formal sector. Evidence from the informal sector supports the idea that greater import competition is associated with sub-contracting of manufacturing activity to the informal sector. We thereby underscore the interaction between trade and labour market institutions in determining the fragmentation of production activity.

References

Acemoglu D., Johnson, S., Mitton, T. 2009. Determinants of Vertical Integration: Financial Development and Contracting Costs. *The Journal of Finance* LXIV (3): 1251–1290

Acemoglu D., Griffith, R., Aghion, P., Zilibotti, F. 2010. Vertical Integration and Technology: Theory and Evidence. *Journal of the European Economic Association* 8 (5): 989–1033

Acemoglu, D., Autor, D., Dorn, D., Hanson, G., Price, B. 2016. Import Competition and the Great US Employment Sag of the 2000s. *Journal of Labor Economics* 34 (S1): S141–S198

Adhvaryu, A., Chari, A., Sharma, S. 2013. Firing Costs and Flexibility: Evidence from Firms' Employment Responses to Shocks in India. *The Review of Economics and Statistics* 95 (3): 725–740

Aghion, P., Burgess, R., Redding, S., Zilibotti, F. 2005. Entry Liberalization and Inequality in Industrial Performance. *Journal of the European Economic Association* 3 (2-3): 291–302

 -. 2008. The Unequal Effects of Liberalization: Evidence from Dismantling the License Raj in India. American Economic Review 98 (4): 1397–1412

Ahsan, A., Pages, C. 2008. Are All Labor Regulations Equal? Evidence from Indian Manufacturing. *IZA Discussion Paper No. 3394*

Alfaro, L., Antras, P., Chor, D., Conconi. P. 2019. Internalizing Global Value Chains: A firm level Analysis. *Journal of Political Economy* 127 (2): 508–559

Alfaro, L., Conconi, P., Fadinger, H., Newman, A. 2016. Do Prices Determine Vertical Integration? The Review of Economic Studies 83 (1): 1–35

Antras, P., Helpman, E. 2004. Global Sourcing. Journal of Political Economy 112 (3): 552–580
Autor, D., Dorn, D., Hanson, G. 2013. The China Cyndrome: Local Labor Market Effects of
Import Competition in the United States. American Economic Review 103 (6): 2121–2168

Banerjee, A., Duflo, E. 2005. Growth Theory through the Lens of Development Economics. In Handbook of Economic Growth, Vol. 1A, edited by Philippe Aghion and Steven Durlauf, 473–552. Amsterdam: North-Holland

Bernard, A., Jensen, B., Schott, P. 2006. Survival of the Best Fit: Exposure to Low-wage Countries and the (uneven) Growth of U.S. Manufacturing Plants. *Journal of International Economics* 68 (1): 219–237

Bernard, A., Fort, T., Smeets, V., Warzynski, F. 2018. Heterogeneous Globalization: Offshoring and Reorganization. *Mimeo*

Besley, T., Burgess, R. 2004. Can Labor Regulation Hinder Economic Performance? Evidence from India. *The Quarterly Journal of Economics* 119 (1): 91-134

Bhattacharjea, A. 2006. Labour Market Regulation and Industrial Performance in India–A Critical Review of the Empirical Evidence. *Working Paper 141*, Centre for Development Economics,

Delhi School of Economics

- Bloom, N., Draca, M., Van Reenen, J. 2016. Technical Change: The Impact of Chinese Imports on Innovation, Diffusion and Productivity. *The Review of Economic Studies* 83 (1): 87–117
- Boehm, J., Oberfield, E. 2018. Misallocation in the Market for Inputs: Enforcement and the Organization of Production. *Mimeo.* Princeton University

Buehler, S., Burghardt, D. 2015. Globalization and Vertical Firm Structure: An Empirical Investigation. *Mimeo.* University of St. Gallen

Burgess, R., Pande, R. 2005. Do Rural Banks Matter? Evidence from the Indian Social Banking Experiment. *American Economic Review* 95 (3): 780–795

Chakraborty, P., Henry, M. 2019. Chinese Competition and Product Variety of Indian Firms. Journal of Comparative Economics 47 (2): 367–395

Charmes, J. 2012. The Informal Economy Worldwide: Trends and Characteristics. *Margin*— The Journal of Applied Economic Research 6 (2): 103–132

Chaurey, R. 2015. Labor Regulations and Contract Labor Use: Evidence from Indian Firms. Journal of Development Economics 114: 224–232

Chen, Y., Ishikawa, J., Yue, Z. 2004. Trade Liberalization and Strategic Outsourcing. *Journal* of International Economics 63: 419–436

Chongvilaivan, A., Hur, J. 2012. Trade Openness and Vertical Integration: Evidence from the U.S. Manufacturing Sector. *Southern Economic Journal* Vol 78 (4): 1242–1264

De Loecker, J. , Goldberg, P. K., Khandelwal, A. K., Pavcnik, N., 2016. Prices, Markups, and Trade Reform. *Econometrica* 84 (2): 445–510

Feenstra, R. 1998. Integration of Trade and Disintegration of Production in the Global Economy. Journal of Economic Perspectives 12 (Autumn): 31–50

Feenstra, R., Hanson, G. 1999. The Impact of Outsourcing and High-Technology Capital on Wages: Estimates For the United States, 1979–1990. *The Quarterly Journal of Economics* 114 (3): 907–940

Fort, T., 2017. Technology and Production Fragmentation: Domestic versus Foreign Sourcing. The Review of Economic Studies 84 (2): 650–687

di Giovanni, J., Levchenko, A., Zhang, J. 2014. The Global Welfare Impact of China: Trade Integration and Technological Change. *Amercan Economic Journal: Macroeconomics* 6 (3): 153– 183

Goldberg P., Pavcnik, N., 2003. The Response of the Informal Sector to Trade Liberalization. Journal of Development Economics 72: 463–496

Goldberg, P., Khandelwal, A., Pavcnik, N., Topalova., P. 2010. Imported Intermediate Inputs

and Domestic Product Growth: Evidence from India. *The Quarterly Journal of Economics* 125 (4): 1727–1767

Gollin, D. 2002. Getting Income Shares Right. Journal of Political Economy 110 (2): 458–74

 -. 2008. Nobody's Business But My Own: Self-Employment and Small Enterprise in Economic Development. Journal of Monetary Economics 55 (2): 219–33

Grossman, G. M., Helpman, E., 2002. Integration versus Outsourcing in Industry Equilibrium. The Quarterly Journal of Economics 117 (1): 85–120.

-, 2004. Managerial Incentives and the International Organization of Production. Journal of International Economics 63 (2): 237–262

-, 2005. Outsourcing in a Global Economy. The Review of Economic Studies 72 (1), 135–159

Gupta, P., Hasan R., Kumar, U., 2008. Big Reforms but Small Payoffs: Explaining the Weak Record of Growth in Indian Manufacturing. in Bery, S, B Bosworth and A Panagariya (eds.), *India Policy Forum*, Vol. 5, Sage, Delhi

Harrigan, J., Reshef, A., Toubal, F. 2016. The March of the Techies: Technology, Trade, and Job Polarization in France, 1994-2007. *NBER Working Paper 22110*. NBER, MA

Hart, O., Holmstrom, B. 2010. A Theory of Firm Scope. *The Quarterly Journal of Economics* 125 (2): 483–513

Hortacsu, A., Syverson, C. 2007. Cementing Relationships: Vertical Integration, Foreclosure, Productivity, and Prices. *Journal of Political Economy* 115 (2): 250–301

Hsieh, C-T., Klenow., P. 2009. Misallocation and Manufacturing TFP in China and India. *The Quarterly Journal of Economics* 124 (4): 1403–1448

Hummels, D., Jorgensen, R., Munch, J., Xiang, C., 2014. The Wage Effects of Offshoring: Evidence from Danish Matched Worker-Firm Data. *American Economic Review* 104 (6): 1597– 1629

Iacovone, L., Rauch, F., Winters, A. 2013. Trade as an Engine of Creative Destruction: Mexican Experience with Chinese Competition. *Journal of International Economics* 89 (2): 379–392

International Labour Organisation (ILO). 2018. Women and Men in the Informal Economy –

A Statistical Picture (Third edition), ILO, Geneva

Jayachandran, S., 2006. Selling Labor Low: Wage Responses to Productivity Shocks in Developing Countries. *Journal of Political Economy* 114 (3): 538–575

Krugman, P. 2008. Trade and Wages, Reconsidered. *Brookings Papers on Economic Activity* 39: 103–154

La Porta, R., Shleifer, A. 2008. The Unofficial Economy and Economic Development. *Brooking* Papers on Economic Activity (Fall): 275–352 -. 2014. Informality and Development. Journal of Economic Perspectives 28 (3): 109–26

Legros, P., Newman, A. 2013. A Price Theory of Vertical and Lateral Integration. *The Quarterly Journal of Economics* 128 (2): 725–770

-. 2014. Contracts, Ownership, and Industrial Organization: Past and Future. *The Journal of Law, Economics, and Organization* 30 (1): i82–i117

Levinsohn, J., Petrin, A. 2003. Estimating Production Functions Using Inputs to Control for Unobservables. *The Review of Economic Studies* 70 (2): 317–341

Liu, R. 2010. Import Competition and Firm Refocusing. *Canadian Journal of Economics* 43 (2): 440–466

Liu, R., Rosell, C. 2013. Import Competition, Multiproduct firms, and Basic Innovation. *Journal of International Economics.* 91 (2): 220–234

Liu, Q., Qiu, L., Zhan, C. 2019. Trade Liberalization and Domestic Vertical Integration: Evidence from China. *Journal of International Economics*. 121: 1–17

Lommerud K. E., Meland F., Straume, O. R., 2009. Can Deunionization Lead to International Outsourcing? *Journal of International Economics* 77 (1): 109-119

Malik, P. 2007. *Labor and Industrial Laws*. Second Edition. Eastern Book Company, Lucknow, India.

Martin, J., Meajean, I. 2014. Low-wage Country Competition and the Quality Content of High-wage Country Exports. *Journal of International Economics* 93 (1): 140–152

McCaig, B., Pavcnik, N. 2018. Export Markets and Labor Allocation in a Low-Income Country. American Economic Review 108 (7): 1899–1941

McGowan, D. 2017. Digging Deep to Compete: Vertical Integration, Product Market Competition and Prices. *The Journal of Industrial Economics* LXV (4): 683–718

McLaren, J., 2000. Globalization and Vertical Structure. *American Economic Review* 90 (5): 1239–1254

-. 2003. Trade and Market Thickness: Effects on Organizations. Journal of the European Economic Association 1 (2/3): 328–336

McMillan, M., Rodrik, D. 2011. Globalization, Structural Change and Productivity Growth. NBER Working Paper 17143, NBER MA

Medina, P. 2017. Import Competition, Quality Upgrading and Exporting: Evidence from the Peruvian Apparel Industry. *Mimeo*. University of Toronto.

Mion, G., Zhu, L. 2013. Import Competition from Offshoring to China: A Curse or Blessing for Firms? *Journal of International Economics* 89 (1): 202–215

Nagaraj, R. 2002. Trade and Labour Market Linkages in India: Evidence and Issues. Economics
Study Area Working Papers 50. East-West Center, Economics Study Area.

Nataraj, S. 2011. The Impact of Trade Liberalization on Productivity: Evidence from India's Formal and Informal Manufacturing Sectors. *Journal of International Economics* 85 (2): 292–301

Ornelas, E., Turner, J. 2008. Trade Liberalization, Outsourcing, and the Hold-up Problem. Journal of International Economics 74 (1): 225–241

-. 2011. Protection and International Sourcing. The Economic Journal 122 (559): 26-63

Restuccia, D., Rogerson., R. 2008. Policy Distortions and Aggregate Productivity with Heterogeneous Establishments. *Review of Economic Dynamics* 11 (4): 707–20

Rodriguez-Lopez, A. 2014. What Drives Offshoring Decisions? Selection and Escape-Competition Mechanisms. *Mimeo*. University of California, Irvine

Silva, J. M. C. Santos, Tenreyro, S. 2006. The Log of Gravity. *The Review of Economics and Statistics* 88 (4): 641–658

Singh, J. 2012. Impact of Surge of Chinese Import on Indian Manufacturing Sector. *Mimeo.* Department of Industrial Policy and Promotion. Ministry of Commerce and Industry, Govt. of India, New Delhi, India.

Stiebale, J., Vencappa, D., 2018. Import Competition and Vertical Integration: Evidence from India. *DICE Discussion Papers 293*, University of Dusseldorf

Sundaram A., Mehta A., Hasan, R., 2017. The Effects of Labor Regulation on Firms and Exports: Theory and Evidence from Indian Apparel. SSRN Working Paper 3077251

Tybout, J. 2000. Manufacturing Firms in Developing Countries: How Well Do They Do, and Why? *Journal of Economic Literature* 38 (1): 11–44

-. 2014. The Missing Middle: Correspondence. *Journal of Economic Perspectives* 28 (4): 235–36

Ulyssea, G. 2010. Regulation of Entry, Labor Market Institutions and the Informal Sector. Journal of Development Economics 91 (1): 87–99

Ulyssea, 2018: Firms, Informality, and Development: Theory and Evidence from Brazil. American Economic Review 108 (8): 2015–2047

Utar, H., Torres-Ruiz, B. 2013. International Competition and Industrial Evolution: Evidence from the Impact of Chinese Competition on Mexican Maquiladoras. *Journal of Development Economics* 105 (1): 267–287



Figure 1: Outsourcing of Manufacturing Jobs, Indian Manufacturing Firms, 1995-2007

Notes: Panel A plots the share of outsourcing expenses on manufacturing jobs in total expenses. Panel B plots percentage of firms involved in outsourcing.



Figure 2: Share of Manufacturing Imports and Import Penetration Ratio for India from China, Indian Manufacturing Industries, 1995-2007

Notes: The line to the left represents average manufacturing imports from China as a share of total manufacturing imports. The line to the right represents the average of the import penetration ratio.



Figure 3: Normalized Expenditure of Outsourcing of Manufacturing Jobs, Indian Manufacturing Firms, 1995-2007

Notes: Figure plots the normalized share of outsourcing expenses of manufacturing jobs in total expenses. 'States with Pro-employer Labour Laws': Andhra Pradesh, Karnataka, Rajasthan, Tamil Nadu and Uttar Pradesh. 'States with Pro-worker Labour Laws': Assam, Bihar, Gujarat, Haryana, Kerela, Madhya Pradesh, Maharastra, Orissa, Punjab, and West Bengal.



Figure 4A: Outsourcing as a function of the elasticity of demand σ by productivity Notes: Low, Medium and High productivity indicate $\gamma = 3, 12, 15$. $\Gamma = 1; w = 5; c = 4.G(k) = \frac{\exp(k)-1}{1-k}$



Figure 4B: Price as a function of the elasticity of demand σ by productivity Notes: Low, Medium and High productivity indicate $\gamma = 3, 12, 15$. $\Gamma = 1; w = 5; c = 4.G(k) = \frac{\exp(k)-1}{1-k}$



Figure 5A: Outsourcing as a function of the elasticity of demand σ by labour law Notes: "Pro-employer" and "Pro-worker" states indicate $w = 4.5, 5.5. \ \gamma = 12;$ $\Gamma = 1; c = 4.G(k) = \frac{\exp(k) - 1}{1 - k}$



Figure 5B: Price as a function of the elasticity of demand σ by labour law Notes: "Pro-employer" and "Pro-worker" states indicate w = 4.5, 5.5. $\gamma = 12; \Gamma = 1; c = 4.G(k) = \frac{\exp(k) - 1}{1 - k}$

-	1 <i>abic</i> 1 . 1 <i>i</i>	Idia b Llade	with Onna an	u Officia		
	Trade wi	th China	Impor	ts from Otl	her Countr	ies
	Imports from China	Exports to China	ASEAN excluding China	US	EU27	World
1992	2.32	2.60	18.95	38.27	124.42	402.50
2001	20.51	10.35	48.88	36.21	116.11	568.70
2007	218.80	84.51	187.24	126.48	288.42	1946.65
Growth (1992-2007)	9339.34%	3150.38%	888.07%	230.49%	131.81%	383.64%
		,	-			

Table 1: India's Trade with China and Others

Notes: Numbers represent real trade values (deflated using Wholesale Price Index of the entire manufacturing sector in India) in INR Millions. Source: Chakraborty and Henry (2019).

		M	Outsourcing	bs	
	Mean	Median	Std. Dev.	Min	Max
Panel A: Firms	with Repo	orted Outs	ourcing Exp	enditure	
Sales	2624.44	257.8	34441.31	0.1	2000000
Assets	2569.80	309.65	24727.86	0.2	1200000
GVA	1404.75	121.6	20711.66	0	1200000
Productivity	0.557	0.496	0.355	0.0001	5.50
Exports	406.27	4.9	5828.86	0	585313
Imports	700.46	7.3	15583.72	0	972704
R&D Intensity	0.013	0.009	0.724	0	89.86
Capital Employed/GVA	7.08	1.73	121.48	0	16789
MCom/TComp	0.062	0.032	0.085	0	1
MIncentives/TIncentives	0.049	0	0.192	0	1
Panel B: Firms u	with No Re	ported Ou	tsourcing E	xpenditur	e
Sales	1640.03	321.9	14519.2	0	1000000
Assets	1616.59	224	9104.12	0.1	347562
GVA	314.55	0	5671.78	0	591644
Productivity	0.533	0.475	0.348	0.0001	4.52
Exports	59.47	0	903.33	0	119211
Imports	117.48	0	3115.78	0	391216
R&D Intensity	0.002	0	0.089	0	18.73
Capital Employed/GVA	3.40	0	81.34	0	10688
MCom/TComp	0.020	0	0.080	0	1
MIncentives/TIncentives	0.010	0	0.085	0	1

Table 2: Firms Reporting Outsourcing of Manufacturing Jobs Vs. Firms Not Reporting Outsourcing of Manufacturing Jobs

Notes: All the numbers reported are in INR Millions. Panel A (B) covers firms that reported positive (zero) expenditure on outsourcing of manufacturing jobs. 'Sales' is the total sales (exports plus domestic sales) of a firm. 'Assets' is the total assets of a firm. 'GVA' is the gross value-added defined as total sales minus total raw material expenditure. 'Productivity' is measured through Levinshon-Petrin (2003) methodology. 'Exports', 'Imports' are the total exports, imports of a firm, respectively. 'R&D intensity' is the GVA share of R&D expenditure. 'Capital Employed' is the amount of capital employed. 'MComp/TComp' is the share of managerial compensation. 'MIncentives/TIncentives' is the share of managerial incentives. For further information on variables see data **Appendix A**.

	Ν	Outsou Ianufactur	rcing ring Jobs
	Total	Share	% of Firms
Panel A			
Aggregate	37.00	0.47	13.86
Panel B: Dividing into State.	s by Lab	our Lau	<i>vs</i>
States with pro-employer Labour Laws	32.46	0.43	11.80
States with Infexible Labour Laws	41.02	0.57	15.47

 Table 3: Outsourcing of Manufacturing Jobs - Total Expenditure, Share of Expenses, Percentage of Firms

Notes: Column (1) calculates the mean outsourcing expenditure by an Indian manufacturing firm. It is expressed in INR Million. Column (2) represents the mean share of outsourcing expenditure in total expenditure of a firm multiplied by 100. Column (3) represents mean percentage of firms involved in outsourcing of manufacturing jobs. 'States with Pro-employer Labour Laws' are: Andhra Pradesh, Karnataka, Rajasthan, Tamil Nadu and Uttar Pradesh. 'States with Pro-worker Labour Laws' are: Gujarat, Maharastra, Orissa, and West Bengal.

		~	þ	~		
Industry Name			Outsou Manufactu	urcing tring Jobs		
	Tot	tal	Sha	are	1 Jo %	irms
	States with Pro-employer Labour Laws	States with Pro-worker Labour Laws	States with Pro-employer Labour Laws	States with Pro-worker Labour Laws	States with Pro-employer Labour Laws	States with Pro-worker Labour Laws
	(1)	(2)	(3)	(4)	(5)	(9)
Final Goods	31.99	42.57	0.51	0.60	12.69	15.22
Intermediate Goods	33.06	39.93	0.35	0.44	10.83	15.65
Notes: Number Durables and Com	s represent average acr sumer Non-Durables, v	oss manufacturing fir vhereas, Intermediate	ms belonging to each 1 Goods include Basic,	user-based industry. F Intermediate and Ca _I	⁷ inal Goods include Co bital goods. Columns (nsumer 1) and (2)
calculate the mean	outsourcing expenditu	re by an Indian manu	ufacturing firm. It is ex	spressed in INR Millio	on. Columns (3) and (4)) represent

\mathbf{S}
Jav
н г
on
ab
<u> </u>
anc
ŝ
trie
ns
nd
F
Se
-ba
ër-
$\mathbf{C}_{\mathbf{S}}$
Ň
Щ
\mathbf{bs} :
Jo
<u>ത</u>
ΞĿ
tu
fac
nu
Ia.
Ę
0
ity
SUS
lte
Γ
hd
e a
nr.
lit
ЭПС
çp
Ĥ
<u>1</u> 0
Ċi.
m
(J
Ę
Dutse
: Outse
e 4: Outse
ble 4: Outse

the mean share of outsourcing expenditure in total expenditure of a firm multiplied by 100. Columns (5) and (6) represent mean percentage of firms involved in outsourcing of manufacturing jobs. 'States with Pro-employer Labour Laws' are: Andhra Pradesh, Karnataka, Rajasthan, Tamil Nadu and Uttar Pradesh. 'States with Pro-worker Labour Laws' are: Gujarat, Maharastra, Orissa, and West Bengal.

Industry Code	Industry Name	М	Outsour	cing 1g Jobs
$\overline{\mathrm{NIC}_{2\text{-digit}}^{2004}}$		Total	Share	% of Firms
15	Foods Products and Beverages	35.50	0.17	7.30
16	Tobacco Products	77.36	1.33	18.01
17	Textiles	29.70	0.73	17.91
18	Wearing Apparel	66.54	1.41	16.17
19	Leather	25.15	1.02	15.19
20	Wood and Wood Products	3.27	0.08	7.20
21	Paper and Paper Products	9.68	0.20	9.33
22	Recorded Media	10.43	1.00	6.74
23	Coke, Refined Petroleum, Nuclear Fuel	257.13	0.15	8.06
24	Chemical and Chemical Products	26.71	0.25	12.88
25	Rubber and Plastics	16.66	0.44	17.37
26	Non-metallic Mineral Products	17.28	0.25	6.68
27	Basic Metals	59.02	0.37	14.63
28	Fabricated Metal Products	35.58	0.88	21.08
29	Machinery and Equipment	35.34	0.82	19.67
30	Office, Accounting & Computing Machinery	1.84	0.02	3.12
31	Electrical Machinery and Apparatus	20.14	0.40	13.33
32	Communication Equipment	6.24	0.25	12.06
33	Medical, Precision and Optical Instruments	10.15	0.53	14.67
34	Motor vehicles, Trailers and Semi-Trailers	1370.55	0.09	6.53
35	Other transport equipment	44.76	0.94	19.54
36	Furniture; Manufacturing n.e.c	64.69	0.72	18.07

Table 5: Outsourcing of Manufacturing Jobs - Total Expenditure, Share of Expenses, Percentage of Firms: At Industry-level (NIC 2-digit)

Notes: Column (1) calculates the mean outsourcing expenditure by an Indian manufacturing firm. It is expressed in INR Million. Column (2) represents the mean share of outsourcing expenditure in total expenditure of a firm multiplied by 100. Column (3) represents mean percentage of firms involved in outsourcing of manufacturing jobs.

	Outsourcing	No of Industries
	Share (%)	
	(1)	(2)
1992 - 2001	$0\!-\!0.25$	67
	0.26 – 0.5	22
	0.6 - 1	7
	$\rangle 1$	4
2002 - 2007	$0\!-\!0.25$	22
	0.26 – 0.5	21
	0.6 - 1	19
	$\rangle 1$	39

Table 6: Distribution of Industries by Outsourcing Share of Manufacturing Jobs

Notes: Column (1) represents the mean outsourcing share of an industry at NIC 4-digit level. Outsourcing Share is defined as the share of outsourcing expenditure in total expenses multiplied by 100. Column (2) counts the number of industries within the relevant range of outsourcing share.

		${ m Year}_{ m 1995-2001}$		(14)	$\begin{array}{c} 0.041 \\ (0.051) \end{array}$	-0.007 (0.006)	0.004 (0.003)	(0.0003)				Yes	0.54	18,136	Yes	Yes	Yes	0 N	No N											
			FLogit	(13)	$\begin{array}{c} 0.016^{***} \\ (0.002) \end{array}$							$\mathbf{Y}_{\mathbf{es}}$	n/a	46,163	Yes	Yes	Yes	0 N O	No		n	Ţ	le	US,	0	fter		ors		
			${ m LR}_{(2013)}$	(12)	0.335^{**} (0.135)	-0.005° (0.003)	0.002 (0.002)	0.0002^{**}				$\mathbf{Y}_{\mathbf{es}}$	0.56	41,515	Yes	Yes	Yes	0 N	No	t variable	imports i	ariff, and	i_{jt-1}^{ina} , is th	inations ¹	(t - 2001)T	e trend a	of $R\&D$	ndard err	iffcance,	
S			${ m TB}_{ m Analysis}$	(11)						-0.013^{*}	0.022^{**} (0.009)	Yes	0.57	41,515	Yes	Yes	${ m Yes}$	0 N O	Yes	depender	Chinese	idia. ' <i>Inpl</i>	$^{, FComp_{IN}^{Ch}}$	here dest	ountries.	fixed tim	ion (sum	trms. Sta	vel of sign	
k Result			FComp UEA	(10)	$\begin{array}{c} 0.016^{***} \\ (0.004) \end{array}$	-0.005 (0.003)	0.001 (0.002)	0.0003^{*}				$\mathbf{Y}\mathbf{es}$	0.55	39,723	Yes	Yes	Yes	N O	γ_{es}	es as the	e share of	994 for In	git level.	in (10); w	v-wage co	aptures a	gy adopt	in real te	nd 1% lev	
enchmar	tring Jobs)/		Long Diff	(6)	0.006^{***} (0.002)	-0.031 (0.038)	0.021 (0.035)	0.0004				$\mathbf{Y}_{\mathbf{es}}$	0.83	7,147	Yes	0 1	Y es M e	0 N	No	al expense	red as the	try j in 16	NIC 4-di	for colum	other lov	T' only ca	l technolo	t-1 and)%, 5% a i	
Jobs: B	s (Manufactu Expenses		First Diff	(8)	0.002^{**} (0.001)	0.0001 (0.002)	0.0001 (0.001)	-0.00004				$\mathbf{Y}\mathbf{es}$	0.01	38,072	No	N0	Y es M	0 N	No	are of tot	is calculat	for indust	s at 2004), except	s from all	2002 - 2007)	ssets) and	at period	lenotes 1(
acturing	ing Expenses Total	${ m Year}_{ m 1995-2007}$		(2)	0.019^{***} (0.004)	-0.003 (0.003)	0.001 (0.002)	0.0002^{*}				\mathbf{Yes}	0.58	35,489	Yes	Yes	Yes	0 N	Yes	s as a she	India. It i	exports	industries	tion (US)	of imports	vhereas '	n, size (as	are used	* ** *** C	
of Manuf	Outsourc			(9)	$0.017^{***}_{(0.005)}$	-0.004 (0.003)	0.001 (0.002)		-0.003			Y_{es}	0.55	39,466	Yes	Yes	Y es	0N0	No	uring job	arket of]	ports and	y Indian	rt destina	he share a	in 2001; v	d of a firr	doption'	reported.	ely.
ourcing e				(5)	0.013^{**} (0.005)	-0.005 (0.003)	$0.002 \\ (0.002)$	0.0002^{**}				$\mathbf{Y}_{\mathbf{es}}$	0.56	41,515	Yes	Yes	Yes	N O N	No	manufact	omestic m	iction, im	fs faced b	ı an expo	$_{-1}^{LWC}$, is t	ne WTO	ge square	mology A	s are not	respectiv
nd Outs				(4)	0.016^{***} (0.005)	-0.005 (0.003)	0.001 (0.002)	~				$\mathbf{Y}_{\mathbf{es}}$	0.56	41,821	Yes	Yes	${ m Yes}$	N O N	No	urcing of	in the de	stic produ	tput tarif	n firms ir	$Comp_{IN,jt}^{Other}$	joining th	de age, ag	and 'Tech	Intercept	
etition a				(3)	$\begin{array}{c} 0.010^{*} \\ (0.006) \end{array}$							$\mathbf{Y}_{\mathbf{es}}$	0.56	41,579	Yes	Yes	NO	Vac	No	on outso	cion ratio	cal domes	t and out	by India	larket. D	of China	ols' inclu	'Assets'	ry level.	
t Comp				(2)	0.009^{**} (0.004)							$\mathbf{Y}_{\mathbf{es}}$	0.56	41,579	Yes	Yes	N O	I es	No	enditure	t pentrat	ed by tot	n of inpu	on faced	export m	t-trends	m Contr	r). Both	ie indust	
: Impor				(1)	$0.013^{***}(0.004)$							$\mathbf{Y}_{\mathbf{es}}$	0.56	41,821	Yes	Yes	Yes	0 N	No	use exp	se impor	lia divid	ogarithn	ompetiti	mbined e	and post	001. 'Fir	Transfe	red at th	
Table 7.					$DComp_{IN,jt-1}^{China}$	$InpTariff_{jt-1}$	$OutTariff_{jt-1}$	$FCom p_{IN,jt-1}^{China}$	$DComp_{IN,jt-1}^{Other\ LWC}$	$DComp_{IN,jt-1}^{China} \times (t - 2001)T$	$DComp_{IN,jt-1}^{China} imes (2002 - 2007)T$	Firm $Controls_{t-1}$	R-Square	N	Firm FE	Year FE	Industry FE (4-digit)*Year Trend I-4	Industry F.E. (2-digit) Tear F.E. Ladacture ETC (2 digit)*Veen ETC	State FE*Year FE	Notes: Columns $(1) - (14)$	$D_{Comp_{IN,jt-1}^{China}}$, is the Chine.	industry j at time t by Inc	<i>OutTariff</i> , are the natural l	measure of Chinese import c	EU, and ASEAN are a co	captures the differential pre-	China joined WTO in 20	expenditure and Technology	in parentheses are cluste	

	Outsou	urcing Exp	enses (Mar	nufacturing	$\rm Jobs)/$
		V	Total Expense	es	Veen
		1995- 1995-	ear -2007		1995 - 2001
	(1)	(2)	(3)	(4)	(5)
$DComp^{China}_{IN,jt-1}$	0.024^{***} (0.006)	0.050^{**} (0.007)	0.026^{***} (0.007)	0.027^{***} (0.006)	$\begin{array}{c} 0.371 \\ \scriptscriptstyle (0.408) \end{array}$
$FComp_{IN,jt-1}^{China}$	0.0001 (0.0001)	$\begin{array}{c} 0.0001 \\ (0.0001) \end{array}$	$\begin{array}{c} 0.0001 \\ (0.0001) \end{array}$	$\underset{(0.0001)}{0.0001}$	$\begin{array}{c} 0.00004 \\ (0.0003) \end{array}$
$InpTariff_{jt-1}$			-0.005^{*} (0.003)	-0.004 (0.003)	-0.010 (0.007)
$OutTariff_{jt-1}$			$\begin{array}{c} 0.002 \\ (0.002) \end{array}$	$\underset{(0.002)}{0.001}$	$\underset{(0.004)}{0.006}$
Firm $Controls_{t-1}$	Yes	Yes	Yes	Yes	Yes
R-Square	0.04	0.06	0.04	0.05	0.04
Ν	37,844	$37,\!844$	$37,\!844$	$32,\!375$	$16,\!529$
Firm FE	Yes	Yes	Yes	Yes	Yes
Year FE	Yes	Yes	Yes	Yes	Yes
Industry FE (4-digit)*Year Trend	Yes	No	Yes	Yes	Yes
Industry FE (3-digit)*Year FE	No	Yes	No	No	No
State FE*Year FE	No	No	No	Yes	No
	1st Sta	ıge			
		D	$Comp_{IN,,j}^{Chin}$	t-1	
$DComp^{China}_{BIMM,jt-1}$	0.150^{***} (0.014)	0.192^{***} (0.041)	0.151^{***} (0.014)	0.151^{***} (0.013)	0.048^{***} (0.016)
F-Stat	155.06		192.77	. ,	10.49

Table 8: Import Competition and Outsourcing of Manufacturing Jobs: IV Results

Notes: Columns (1) - (5) use expenditure on outsourcing as a share of total expenses as the dependent variable. $DComp_{IN,jt-1}^{China}$ is the Chinese import pentration ratio in the domestic market of India. We use $DComp_{BIMM,jt-1}^{China}$ as the instrument for $DComp_{IN,jt-1}^{China}$. We measure $DComp_{BIMM,jt-1}^{China}$ using imports from other developing countries such as Brazil (B), Indonesia (I), Malaysia (M) and Mexico (M). InpTariff and OutTariff are the natural logarithm of input and output tariffs faced by Indian industries at 2004 NIC 4-digit level. $FComp_{IN,jt-1}^{China}$ is the measure of Chinese import competition faced by Indian firms in an export destination (US). 'Firm Controls' include age, age squared of a firm, size (assets) and technology adoption (sum of R&D expenditure and Technology Transfer). Both 'Assets' and 'Technology Adoption' are used at period t-1 and in real terms. Standard errors in parentheses are

clustered at the industry level. Intercepts are not reported. *,**,*** denotes 10%, 5% and 1% level of significance, respectively.

Table 9: Import Competition and Outsourcing of Manufacturing Jobs: Controlling for Import Competition from Other Regions and Intermediate Inputs

			Outsourcing	g Expenses	(Manufactur	ing Jobs)/			
	In	port Compe	etition from	Other Reg	ons	Case of	Intermediat	e Inputs	
	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)	
$DComp_{IN,jt-1}^{World}$	-0.0007 (0.001)								
$DComp_{IN, Jt-1}^{China}$	0.013^{***}	$0.011^{***}_{(0.003)}$	$0.010^{***}_{(0.003)}$	$0.008^{***}_{(0.003)}$	$0.009^{***}_{(0.003)}$	$0.016^{***}_{(0.004)}$	$0.013^{***}_{(0.004)}$	$0.011^{**}_{(0.004)}$	
$DComp_{IN,jt-1}^{High-Income}$	~	-0.0003	~	~	~	~	~	~	
$DComp_{IN,jt-1}^{NA}$		~	-0.041^{**}		-0.040^{**}				
$DComp_{IN,jt-1}^{EU}$			0.014		0.010				
$DComp_{IN}^{LA},_{jt-1}$			(2222)	-0.010	-0.005				
$DComp_{IN,jt-1}^{LDC}$				-0.009	-0.00002				
$DComp_{IN,jt-1}^{MENA}$				$(0.00)^{(0.00)}$	-0.005				
$DComp_{IN,jt-1}^{SA}$				(0.032)	(0.033)				
$InpDComp_{IN,jt-1}^{China}$				(ocn.n)	(ee0.0)	-0.003	0.004	0.003	
$InpTariff_{jt-1}$						(0000)	-0.005^{*}	-0.006^{*}	
$Out Tariff_{jt-1}$							0.002	(0.002)	
$FComp_{IN,jt-1}^{China}$								0.0001	
Firm $Controls_{t-1}$	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	
R-Square	0.56	0.56	0.56	0.56	0.56	0.59	0.59	0.59	
Z	38,625	38,625	38,625	38,625	38,852	38,131	38,131	37,844	
Firm FE Indications (P. discit)&Wroom EE	${ m Yes}_{ m voc}$	${ m Yes}$	${ m Yes}$	${ m Yes}$	${ m Yes}$	${ m Yes}$	${ m Yes}$	${ m Yes}$	
Notes: Columns (1) - (8) use expenditure	e on outso	urcing as a	res share of to	res otal expens	es as the de	$\frac{1}{1}$	res /ariable. 'L	1es $Comp_{IN,it-a}^{China}$	$_{1}$, is the
Chinese import pentration ratio in the dom-	estic marl	set of India	. It is calc	ulated as t	ne share of	Chinese ir	nports in ii	ndustry j a	time t by
India divided by total domestic productio	m, import	s and expo	rts for indu	istry j in 1	994 for Ind	ia. ' <i>DCom</i>	$vp_{IN,jt-1}^{World}$,	$DComp_{IN,ji}^{High}$	$^{-Income},$
$, DComp_{IN,jt-1}^{NA}, , DComp_{IN,jt-1}^{EU}, , DComp_{IN}^{LA},$	$(j_{t-1}), DC$	$Comp_{IN,jt-1}^{LDC}$, , DComp	$_{tN,jt-1}^{MENA}$, D	$Comp^{SA}_{IN,jt-1}$	' are imp	ort penetra	tion indices	in case of
World $(World)$, High-Income countries (C	ECD plu	s non-OEC	(D) (High -	- $Income$),	North Ame	rica (NA)	, European	Union $(Et$	'), Latin
America (LA) , Least Developed Countries	(LDC), N	fiddle East	and North	ı African c	ountries (M)	ENA, an	d South A	sia (SA) res	pectively.
$(InpDComp_{IN,it-1}^{China})$ is a measure of imports	s of intern	nediate inp	uts from C	hina. ' Inp_{1}	<i>ariff</i> and	OutTari	ff' are the	natural log	arithm of

level of significance, respectively.

adoption (sum of R&D expenditure and Technology Transfer). Both 'Assets' and 'Technology Adoption' are used at period t-1 and in real

terms. Standard errors in parentheses are clustered at the industry level. Intercepts are not reported. *,**,*** denotes 10%, 5% and 1%

input and output tariffs faced by Indian industries at 2004 NIC 4-digit level. ' $FComp_{IN,jt-1}^{China}$ ' is the measure of Chinese import competition

faced by Indian firms in an export destination (US). 'Firm Controls' include age, age squared of a firm, size (assets) and technology

				Outsourcing E	xpenses (Manu Total Expenses	facturing Jobs	s)/		
	${ m Skill}_{ m Intensity}$	Factories	Total Factor Productivity	Multi-Product	Managerial Compensation	Managerial ^{Wages}		Manager	ial
					4		Total	Executives	Non-Executives
	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)	(6)
$DComp_{IN,jt-1}^{China}$	0.059^{*}	-0.014	0.339^{**}	0.005	-0.053	-0.036)	$0.007 \\ (0.007)$	0.007	0.464^{***} (0.176)
$DComp_{IN,jt-1}^{China} imes SkIntens_{j,1995-2001}$	0.029		~	~			~	~	~
$DComp_{IN,jt-1}^{China} imes Factories_{j,1995-2001}$	(0-0-0)	0.004							
$DComp_{IN,jt-1}^{China} imes MPFirm$				0.014^{***}					
$DComp_{IN,jt-1}^{China} imes TFP_{j,1995-2001}$			$0.541^{**}_{(0.236)}$						
$DComp_{IN,jt-1}^{China} imes MComp_{j,1995-2001}$					-0.019				
$DComp^{China}_{IN,jt-1} imes ~MW ages_{j,1995-2001}$					(0-0-0)	-0.009			
$DComp_{IN,jt-1}^{China} imes MIncentives_{j,1995-2001}$						(000.0)	-0.001	-0.001	0.040^{***}
Firm Controls $_{t-1}$	Yes	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Y}_{\mathbf{es}}$	Yes	${ m Yes}$	\mathbf{Yes}	Yes	Yes	Yes
R-Square	0.59	0.59	0.59	0.65	0.59	0.59	0.59	0.59	0.62
Z	35,548	35,548	35,548	41,515	35,548	35,548	35,161	35,161	24,660
Firm FE	Yes	$\dot{\mathrm{Yes}}$	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Y}_{\mathbf{es}}$	$\dot{\mathrm{Yes}}$	\dot{Y}_{es}	$\dot{\mathrm{Yes}}$	$\mathbf{Y}_{\mathbf{es}}$
Product FE	N_{0}	N_{O}	N_{O}	N_{O}	N_{O}	N_{O}	N_{O}	N_{O}	No
Year FE	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}
Industry FE (3-digit)*Year FE	Yes	\mathbf{Yes}	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Y}_{\mathbf{es}}$	${\rm Yes}$	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	$\mathbf{Y}_{\mathbf{es}}$
State FE*Year FE	γ_{es}	$_{\rm Yes}$	\mathbf{Yes}	\mathbf{Yes}	${ m Yes}$	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}
Notes: Columns $(1) - (9)$ use exper	nditure o	n outsourci	ng of manufa	cturing jobs as	a share of t	otal expense	es as the c	dependent v	ariable.
$DComp_{IN,jt-1}^{China}$ is the Chinese import	t pentrat	ion ratio in	the domestic	c market of Ind	lia. It is calc	ulated as th	te share o	f Chinese im	ports in
industry j at time t by India divided l	by total a	domestic pr	oduction, im	ports and expo	rts for indus	try j in 199.	4 for Indi	a. ' $FComp_{IN}^{Ch}$	$_{it}^{ina}$, is the
measure of Chinese import competiti	ion faced	l by Indian	firms in an e	xport destinati	on (US). Sk	Intens' is a]	proxy for	skill intensit	y at the
industry level. It is defined as the sh	are of no	n-productic	on workers to	total employee	es at the NIC	3-digit leve	el. ' <i>Facto</i>	ries' is the n	umber of
factories at the 3-digit level NIC :	2004. 'T1	FP' is total	factor produ	ctivity at firm	level estimat	ed using Le	vinshon a	nd Petrin (2	(003).
MPFirm' is a dummy variable wh	ich takes	a value 1 v	when a firm p	produces more	than 1 prod	ict. ' $MComp$	is the sl	hare of man	ıgerial
compensation in total labour compens	sation for	: firm i . 'MI	Wages' is the	share of total 1	nanaegrial w	ages in tota	J wages fo	or firm i . 'M	Incentives,
is the share of total managerial incen	tives in t	otal incenti	ves for firm i	. 'Firm Contro	ls' include a	ge, age squa	red of a f	irm, size (as	sets) and
technology adoption (sum of $R\&D ex$)	penditur	e and Techn	nology Transf	fer). Both 'Ass	ets' and 'Tec	thnology Ad	option' ar	re used at po	priod $t-1$
and in real terms. Standard errors in	parenth	eses are clu	stered at the	industry level.	Intercepts a	are not repo	rted. *,**	,*** denotes	10%, 5%
		and 1%	level of signi	ficance, respect	ively.				

Table 11: Import Compe	stition and	Outsourci	ng of Man	ufacturing .	Jobs: 'lestin	g for Labou	: Market I	{egulations
			Outs	ourcing Expe	enses (Manurad Jotal Expenses	turing Jobs//		
			Using	; Gupta et al.	(2009)			Using Adhvaryu et al. (2013)
					Single-Prod	Multi-Prod	$\operatorname{Baseline}_{\operatorname{Character}}$	
	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)
$DComp_{IN,jt-1}^{China}$	$0.020^{***}_{(0.003)}$	$0.015^{***}_{(0.003)}$	$0.025^{***}_{(0.005)}$	$0.018^{***}_{(0.004)}$	$0.004 \\ (0.013)$	$0.019^{***}_{(0.004)}$	$0.046^{***}_{(0.013)}$	0.010
$DComp_{IN,jt-1}^{China} imes LMktR_s$	-0.017^{***}	-0.019^{***}	-0.024^{***}	-0.021^{**}	0.022	-0.024^{**}	-0.037^{**}	
$InpTariff_{jt-1}$	(100.0)	(enn.n)	(100.0)	-0.004	(+00.0)	(010.0)	(10.01)	
$InpTariff_{jt-1} imes LMktR_s$				0.0003 0.0003				
$OutTariff_{jt-1}$				0.001				
$OutTariff_{jt-1} imes LMktR_s$				0.0003				
$FComp_{IN,jt-1}^{China}$				0.0001				
$FComp_{IN,jt-1}^{China} imes LMktR_{s}$				0.0001				
$DComp_{IN,jt-1}^{Other\ LWC}$				(1000.0)				
$DComp_{IN,jt-1}^{Other\ LWC} imes\ LMktR_s$				(0.003) 0.004				
$DComp_{IN,jt-1}^{China} imes pro - wor \ker$				(600.0)				0.046^{**}
$DComp_{IN,jt-1}^{China} imes neutral$								0.011^{**} (0.005)
Firm $Controls_{t-1}$	Yes	\mathbf{Yes}	\mathbf{Yes}	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Y}_{\mathbf{es}}$	\mathbf{Yes}	\mathbf{Yes}	Yes
R-Square	0.59	0.59	0.62	0.58	0.69	0.63	0.62	0.62
Z	38, 131	38, 131	38, 131	36, 135	3,519	37,996	38, 131	32, 349
Firm FE	${\rm Yes}$	${\rm Yes}$	\mathbf{Yes}	${ m Yes}$	\mathbf{Yes}	${\rm Yes}$	${ m Yes}$	${ m Yes}$
Year FE	${\rm Yes}$	${\rm Yes}$	\mathbf{Yes}	${ m Yes}$	\mathbf{Yes}	${ m Yes}$	${ m Yes}$	${ m Yes}$
Industry FE (4-digit)*Year Trend	${\rm Yes}$	N_{O}	N_{O}	N_{O}	N_{O}	N_{O}	No	No
Industry FE (3-digit)*Year FE	No	\mathbf{Yes}	\mathbf{Yes}	${ m Yes}$	\mathbf{Yes}	\mathbf{Yes}	Yes	${ m Yes}$
State FE*Year FE	\mathbf{Yes}	Yes	Yes	\mathbf{Yes}	Yes	Yes	$\mathbf{Y}^{\mathbf{es}}$	Yes
Notes: Columns $(1) - (8)$ use \exp	enditure or	1 outsourcin	g of manufa	teturing jobs	s as a share of	total expens	ies as the d	ependent variable.
$DComp_{IN,jt-1}^{China}$, is the Chinese imp	ort pentrati	ion ratio in	the domesti	c market of	India. It is c	alculated as t	he share of	Chinese imports in
industry j at time t by India divide	ed by total	domestic pr	oduction, ir	nports and e	exports for in	dustry j in 19	994 for Ind	ia. 'InpTariff' and
OutTariff, are the natural logar	ithm of inp	ut and outp	ut tariffs fa	ced by India	an industries	at 2004 NIC ⁴	4-digit. FC	$\mathcal{I}omp_{IN,jt-1}^{China}$, is the
measure of Chinese import competit	tion faced b	y Indian fir	ms in an ex	port destina	tion (US). D	$Comp_{IN,jt-1}^{Other\ LV}$	VC , is the s	hare of imports from
all other low-wage countries. LM	ktR_s ' is an	indicator for	r labour ma	rket regulat:	ion. It takes a	a value 1 if a	state has p	ro-employer labour
market laws and 0 otherwise. 'pro	– <i>worker</i> ' t	akes a value	e 1 if a state	c = Gujarat,	Maharastra,	Orissa, and ¹	West Beng	al. ' <i>neutral</i> ' takes a
value 1 if a state = Assam, Bihar, \mathbf{F}	Iaryana, Ja	mmu and K	ashmir, Pur	njab and Ut	tar Pradesh.	Firm Contro	ls' include	age, age squared of a
firm, size (assets) and technology a	doption (su	m of R&D	expenditure	and Techno	ology Transfer). Both 'Asse	ets' and 'Te	schnology Adoption'

are used at period t-1 and in real terms. Standard errors in parentheses are clustered at the industry level. Intercepts are not reported. *,**,*** denotes 10%, 5% and 1% level of significance, respectively.

	Outsou	rcing Exp	enses (Pr	ofessional	Jobs) / Jace	Outsourcing	
			otal Expension	es	, i i	Intensity	
	(1)	(2)	(3)	(4)	(5)	(9)	
$DComp_{IN,it-1}^{China}$	-0.0004	-0.002	-0.001	-0.003	-0.003	0.014	
	(0.005)	(0.010)	(0.005)	(0.008)	(0.008)	(0.010)	
$DComp_{IN,it-1}^{China} imes LMktR_{s}$	-0.020	-0.019	-0.013	-0.008	-0.007	0.012	
	(0.013)	(0.012)	(0.011)	(0.013)	(0.013)	(0.021)	
$InpTariff_{jt-1}$			0.005	0.005	0.004 (0.012)	0.0001	
InnTariff. + XMk+R			-0.003	-0.004		0 001	
Satisfy the the second			(0.011)	(0.014)	(0.015)	(0.010)	
$OutTariff_{jt-1}$			-0.001	-0.001	-0.001	0.001	
$OutTariff_{i^{\pm}-1} imes LMktR_{c}$			0.005	0.005	0.004	0.002	
Saturday 1-166 from the o			(0.013)	(0.013)	(0.013)	(0.009)	
$FComp_{IN,jt-1}^{China}$				0.0001	0.0001	0.0002	
				(ennn.n)	(ennn-n)	(ennn.n)	
$FComp_{IN,jt-1}^{Chma} \times LMktR_s$				-0.0001	-0.0001	-0.0003^{*}	
$DComp_{TM}^{Other} L^{WC}$					0.006	0.001	
					(0.012)	(600.0)	
$DComp_{IN,jt-1}^{Other \ LWC} imes \ LMktR_s$					-0.007	-0.001	
Firm $Controls_{t-1}$	Yes	Yes	Yes	Yes	Yes	Yes	
R-Square	0.35	0.35	0.35	0.35	0.34	0.45	
N	31,824	31,824	31,824	31,577	30,228	30,442	
Firm FE	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	m Yes	
Year FE	\mathbf{Yes}	\mathbf{Yes}	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Y}_{\mathbf{es}}$	\mathbf{Yes}	${ m Yes}$	
Industry FE (4-digit)*Year Trend	\mathbf{Yes}	N_{O}	N_{O}	N_{O}	No	N_{O}	
Industry FE (3-digit)*Year FE	N_{O}	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	${ m Yes}$	
State FE*Year FE	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	${ m Yes}$	
mms $(1) - (5)$ use expenditure on outso	ourcing on p	rofessional	l jobs as a	share of tot	al expenses	as the dependent	var

Ē ¢ ç \langle ζ 0 E

(assets) and technology adoption (sum of R&D expenditure and Technology Transfer). Both 'Assets' and 'Technology Adoption' are used at Column (6) uses outsourcing intensity (it takes a value 1 when outsourcing expenditure on professional jobs of a firm is greater than zero) as share of Chinese imports in industry j at time t by India divided by total domestic production, imports and exports for industry j in 1994 $DComp_{IN,jt-1}^{Other LWC}$, is the share of imports from all other low-wage countries. $LMktR_s$, is an indicator for labour market regulation. It for India. *'InpTariff'* and 'OutTariff' are the natural logarithm of input and output tariffs faced by Indian industries at 2004 NIC the dependent variable ' $DComp_{IN,jt-1}^{China}$ ' is the Chinese import pentration ratio in the domestic market of India. It is calculated as the takes a value 1 if a state has pro-employer labour market laws and 0 otherwise. 'Firm Controls' include age, age squared of a firm, size period t-1 and in real terms. Standard errors in parentheses are clustered at the industry level. Intercepts are not reported. *, **, *** iable. 4-digit level. $FComp_{IN,jt-1}^{China}$ is the measure of Chinese import competition faced by Indian firms in an export destination (US). Notes: Colur

denotes 10%, 5% and 1% level of significance, respectively.

		a firm is on	Outract or sel	${ m itS}=1$ lis its output to	other enterprises		${ m OutS} = 1$ firm sells to other enterprises	$\begin{array}{c} \mathrm{OutS} = 1 \\ \mathrm{a~firm~is~on~contract} \end{array}$	Out per w	put orker
		Probit	Logit		Urban	Rural			OutS = 1	OutS = 0
	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)	(6)	(10)
$DComp_{IN,jt}^{China}$	$0.044^{**}_{(0.018)}$	$0.057^{***}_{(0.020)}$	$0.061^{***}_{(0.021)}$	$0.077^{***}_{(0.020)}$	$0.048^{**}_{(0.019)}$	$0.155^{***}_{(0.034)}$	$0.078^{***}_{(0.019)}$	$0.031^{**}_{(0.018)}$	$1.363 \\ \scriptstyle (1.156)$	$\begin{array}{c} 0.101 \\ \scriptstyle (0.262) \end{array}$
$DComp_{IN,jt}^{China} imes LMktR_{s}$				-0.076^{***}	-0.045^{***}	-0.156^{***}	-0.087^{***}	-0.013	-1.184^{***}	-0.384
Othon Contucle	Voc	$\mathbf{V}_{\alpha\alpha}$	$\mathbf{V}_{\alpha\alpha}$	(0.016) Vor	(etu.u)	()(0.00)		(etu.u)	V_{06}	Voc
Other Controls	IGS	IGS	IGS	IGS	IGS	IGS	ICS	IGS	IGS	ICS
Firm Controls	\mathbf{Yes}	${ m Yes}$	${ m Yes}$	${\rm Yes}$	${ m Yes}$	${ m Yes}$	${ m Yes}$	${ m Yes}$	${ m Yes}$	${ m Yes}$
R-Square	0.32	0.27	0.28	0.33	0.29	0.37	0.29	0.24	0.79	0.73
Ν	133,939	133,917	133,917	133,939	82,516	51,423	133,916	36,817	101, 167	84,043
Industry FE	\mathbf{Yes}	\mathbf{Yes}	${ m Yes}$	${ m Yes}$	\mathbf{Yes}	${ m Yes}$	${ m Yes}$	Yes	${ m Yes}$	γ_{es}
Industry FE (4-digit)*Year Trend	Yes	$\mathbf{Y}_{\mathbf{es}}$	${ m Yes}$	${ m Yes}$	${ m Yes}$	${ m Yes}$	${ m Yes}$	\mathbf{Yes}	${ m Yes}$	${ m Yes}$
State FE*Year FE	Yes	\mathbf{Yes}	$\mathbf{Y}_{\mathbf{es}}$	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Y}_{\mathbf{es}}$
Notes: Columns $(1) - (6)$ us	se an outs	ourcing ir	dicator va	riable whic	h takes a ve	ulue 1 if a fi	rm sells or is on contra	ict to sell to anot	ther	
printe anterprise or to a contre	actor mic	ldlaman a	+ho dono	adont mania	hla Column	e agan (2) t	n outsourcing indicator	h doidm oldeirer a	toleos o	

Table 13: Import Competition, Labour Market Dynamics, and Outsourcing: Using Data from the Informal Sector

Chinese imports in industry j at time t by India divided by total domestic production, imports and exports for industry j in 1994 for India. value 1 when a firm sells most of its output to another firm. Column (8) uses an outsourcing indicator variable which takes a value 1 if a dependent variable. ' $DComp_{IN,jt-1}^{China}$ ' is the Chinese import pentration ratio in the domestic market of India. It is calculated as the share of $'LMktR_s'$ is an indicator for labour market regulation. It takes a value 1 if a state has pro-employer labour market laws and 0 otherwise. firm is on contract to sell to another firm or middleman. Columns (9) and (10) use logarithm of gross value-added per worker as the ariable which tal SOULCING IIIUICAUU Column (1) uses an our to a contractor/miggieman as the dependent private enterprise or

interaction terms with labour market regulation. 'Firm Controls' include assets (size) and GVA in real terms. Standard errors in parentheses 'Other Controls' include input and output tariffs faced by Indian industries at 2004 NIC 4-digit level, export market competition and their are clustered at the industry level. Intercepts are not reported. *,**,*** denotes 10%, 5% and 1% level of significance, respectively.

	•	•))						
		Cos	sts			Pric	es			Sal	es	
	Outsour	$\operatorname{cing} = 1$	Outsourd	sing = 0	Outsourc	ing = 1	Outsourc	ing = 0	Outsourd	ing = 1	Outsourc	ing = 0
	${ m Pro-work}_{ m States}$	${ m Pro-emp}_{ m States}$	\Pr_{States}	${ m Pro-emp}_{ m States}$	${ m Pro-work}_{ m States}$	${ m Pro-emp}_{ m States}$	\Pr_{States}	Pro-emp States	${ m Pro-work}_{ m States}$	Pro-emp States	${\rm Pro-work}_{{\rm States}}$	Pro-emp States
	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)	(6)	(10)	(11)	(12)
$DComp_{IN,jt-1}^{China}$	-1.180^{**} (0.484)	-0.943^{***} (0.155)	-0.304 (0.326)	-0.211 $_{(0.172)}$	-0.471^{***} (0.138)	-0.408^{**} (0.198)	-0.304 $_{(0.326)}$	-0.211 $_{(0.172)}$	$0.707^{*}_{(0.430)}$	$0.535^{*}_{(0.276)}$	-0.537 (0.633)	0.689 (0.665)
Firm $Controls_{t-1}$	${ m Yes}$	Yes	Yes	Yes	Yes	Yes	Yes	Yes	γ_{es}	\mathbf{Yes}	\mathbf{Yes}	γ_{es}
$\operatorname{R-Square}$	0.69	0.59	0.89	0.89	0.89	0.67	0.89	0.89	0.87	0.84	0.84	0.85
N	10,227	3,303	17,097	8,435	10,227	3,303	17,097	8,435	10,227	3,303	17,097	8,435
Firm-Product FE	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}
Industry FE (3-digit)*Year FE	\mathbf{Yes}	Yes	\mathbf{Yes}	${ m Yes}$	\mathbf{Yes}	Yes	${ m Yes}$	${ m Yes}$	Yes	${ m Yes}$	${\rm Yes}$	$\mathbf{Y}_{\mathbf{es}}$
State FE*Year FE	\mathbf{Yes}	$\mathbf{Y}_{\mathbf{es}}$	\mathbf{Yes}	${ m Yes}$	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Y}_{\mathbf{es}}$	\mathbf{Yes}	${ m Yes}$	$\mathbf{Y}_{\mathbf{es}}$	\mathbf{Yes}	\mathbf{Yes}	$\mathbf{Y}_{\mathbf{es}}$
Notes: Columns $(1) - (4)$ use	e logarithm	of prices, c	(5)	-(8) use	logarithm e	of marginal	costs, and	columns	(9) - (12)	use logarit	hm of	

•	unsina
-	Sch
e F	ž
	ре
	ы Ц
د	2 2 2
	esting
t	
-	Jobs:
•	uring
۔ د	nutact
Ļ	Тaл
۴	4
ح	d
•	ourcing (
-	Juts
(2
	and
•	npetition
ζ	Con
-	lmport
٣	ч е
E	Tabl

Chinese import pentration ratio in the domestic market of India. It is calculated as the share of Chinese imports in industry j at time t by 'Efficiency Gains and Product Market Competition: Evidence from India' by Chakraborty, Henry and Singh (2019). 'DCompEting' is the total sales of a firm as the dependent variable, respectively. The data on costs, prices, and markups have been sourced from the paper

a firm, size (assets) and technology adoption (sum of R&D expenditure and Technology Transfer). Both 'Assets' and 'Technology Adoption' are used at period t-1 and in real terms. Standard errors in parentheses are clustered at the industry level. Intercepts are not reported. *, **, **** denotes 10%, 5% and 1% level of significance, respectively.

India divided by total domestic production, imports and exports for industry j in 1994 for India. 'Firm Controls' include age, age squared of

58

Appendix

A Data

We use an annual panel of Indian manufacturing firms that covers 9000+ firms, across 105 industries, over the period of 1995-2007. Data is used from the PROWESS database of the Centre for Monitoring Indian Economy (CMIE). All monetary-based variables measured in Millions of Indian Rupees (INR), deflated by 2005 industry-specific Wholesale Price Index (WPI). We use 2004 National Industrial Classification (NIC). We use import penetration data from the UN-COMTRADE database.

Variable definitions

Expenditure on Outsourcing of Manufacturing Jobs: These are expenses incurred by firms to get their manufacturing requirements fulfilled from outside parties. It is a normal practice followed by firms to outsource a part of their requirements. Also, certain firms that manufacture large products (like car manufacturers) outsource requirements to outside firms as it may not be feasible or economical for them to manufacture all inputs into the product. Many firms outsource the whole of their manufacturing process and add their brand name to the product. This variables reports any amount expended by a firm on outsourcing of manufacturing jobs. It includes labour charges, fabrication charges, processing charges, machining charges, fettling charges and the like. Other terms include - conversion charges, contracted production and sub-contracted production.

Expenditure on Outsourcing on Professional Jobs: These are the expenses incurred by firms for engaging external professional services. The services include: (i) Software development fees, (ii) IT enabled services charges, (iii) Cost audit fees, (iv) Legal charges, (v) Miscellaneous professional services, (vi) Auditors fees, and (vii) Consultancy fees. Such services exclude those related to manufacturing jobs, selling and distribution, financial intermediaries or financial services.

Outsourcing Indicator (NSSO): It takes a value 1 if a firm sells or is on contract to sell to another private enterprise or to a contractor/middleman. It can be divided into two parts -(1) takes a value 1 when a firm sells most of its output to another firm; and (2) takes a value 1 if a firm is on contract to sell to another firm or middlemen.

Chinese Competition at Domestic Market $(DComp_{IN,jt}^{China})$: This is the Chinese import pentration ratio in the domestic market of India. It is calculated as the share of Chinese imports in industry j at time t by India divided by total domestic production, imports and exports for industry j in 1994 for India.

Imported Intermediate Inputs from China $(InpDComp_{IN,jt-1}^{China})$: This is an index of imported

intermediate inputs from China. We weight the I–O coefficient of each input sector (at NIC 4-digit level) by its import share, and then by the Chinese share in imports for that sector. By summing these measures, we arrive at a measure that gives the average weighted sum of intermediate goods imported from China at a sectoral level, where the weights are given by the coefficients of the I-O table.

States with pro-employer Labour Laws $(LMktR_s)$: This is an indicator for labour market regulation. It takes a value 1 if a state has pro-employer labour market laws and 0 otherwise. States with pro-employer labour laws are: Andhra Pradesh, Karnataka, Rajasthan, Tamil Nadu, and Uttar Pradesh. States with pro-worker labour laws are: Gujarat, Maharastra, Orissa, and West Bengal. Source: Gupta, Hasan and Kumar (2009).

Chinese Competition at Export Market $(FComp_{IN,jt}^{China})$: This is the Chinese import ratio in one Indian export market, namely the US. We also use a combined ratio of the US, EU and ASEAN. It is defined as the share of Chinese imports in total imports.

Import Penetration from Other Low-Wage Countries $(DComp_{IN,jt-1}^{Other LWC})$: This is the import penetration ratio in the domestic market of India from low-wage countries other than China. It is constructed in a manner similar to $DComp_{IN,jt}^{China}$.

Chinese Competition for Other Developing Countries $(DComp_{BIMM,jt-1}^{China})$: We use $DComp_{BIMM,jt-1}^{China}$ as an instrument for $DComp_{IN,jt-1}^{China}$. We measure $DComp_{BIMM,jt-1}^{China}$ using Chinese imports by other developing countries such as Brazil (B), Indonesia (I), Malaysia (M) and Mexico (M).

Import Penetration Ratio from World $(DComp_{IN,jt-1}^{World})$: This is an aggregate import penetration ratio.

Import Penetration Ratio from High-Income Countries $(DComp_{IN,jt-1}^{High-Income})$: This is an import penetration ratio of high-income countries. It includes both OCED and non-OECD countries.

Import Penetration Ratio from North America $(DComp_{IN,jt-1}^{NA})$: This is an import penetration ratio of North America (USA, Canada and Mexico).

Import Penetration Ratio from European Union $(DComp_{IN,jt-1}^{EU})$: This is an import penetration ratio of the 27 European Union countries.

Import Penetration Ratio from Latin America $(DComp_{IN,jt-1}^{LA})$: This is an import penetration ratio of South American countries.

Import Penetration Ratio from Least Developed Countries $(DComp_{IN,jt-1}^{LDC})$: This is an import penetration ratio of Least Developed countries.

Import Penetration Ratio from Middle East and North Africa $(DComp_{IN,jt-1}^{MENA})$: This is an import penetration ratio of Middle East and North African countries.

Import Penetration Ratio from South Asia $(DComp_{IN,jt-1}^{SA})$: This is an import penetration

ratio of South Asian countries.

Input/Output Tariffs: Input/Output tariffs at the 4-digit industry level, obtained from Ahsan and Mitra (2014) for the period of 1990-2003, with the balance collected from the TRAINS-WITS tariff database.

Productivity: firm level Total Factor Productivity (TFP) is computed using the Levinsohn and Petrin (2003) methodology.

Mcomp/Tcomp: The share of managerial compensation in total labour compensation; compensation defined as the sum of all salaries, and additional bonuses.

MWages/TWages: The share of managerial wages in total wages of a firm.

MIncentives/TIncentives: The share of managerial incentives in total incentives of a firm.

Skill intensity: It is defined as the ratio of non-production workers to total employees at the 3-digit level of 2004 NIC. We obtain this from two different sources - for the years 1995-2000, data has been generously shared by Dr. Sangeeta Ghosh; and for 2001-2007 from various publications of the Annual Survey of Industries, Central Statistical Organization, India.

Factories: The number of factories at the 3-digit level of 2004 NIC.

Intermediate goods: Goods classified according to the I-O table as inputs by end-use. It combines intermediates, capital and basic goods.

Final goods: Goods classified according to the I-O table as final products by end-use. It combines consumer durable and consumer non-durable goods.

TechAdop/GVA: Share of R&D expenditure and Royalty Payments for Technical Knowhow in gross value-added.

Cap/GVA: Share of the total amount of capital employed in gross value-added.

GVA: Gross Value-Added = Total Sales - Total Raw Material Expenditure.

Assets: Total assets of a firm.

Sales: Total sales (exports + domestic sales) of a firm.

Exports: Total exports of a firm.

Imports: Total imports (imports of raw materials, finished goods, stores & spares, and capital goods)

Ownership: It indicates whether a firm is domestic-owned or foreign-owned.

Age: Age of a firm in years.

B Tables

Industry Code	Industry Name	Chinese World	Imports/ Imports
$\operatorname{NIC}_{2\text{-digit}} 2004$		1992 - 2001	2002-
		(1)	(2)
15	Foods Products and Beverages	1.72	3.04
16	Tobacco Products	0.69	4.95
17	Textiles	21.66	43.02
18	Wearing Apparel	9.11	18.84
19	Leather	8.80	33.70
20	Wood and Wood Products	2.81	15.73
21	Paper and Paper Products	0.92	5.39
22	Recorded Media	1.37	9.24
23	Coke, Refined Petroleum, Nuclear Fuel	10.05	10.97
24	Chemical and Chemical Products	7.94	20.12
25	Rubber and Plastics	2.27	13.51
26	Non-metallic Mineral Products	2.53	17.32
27	Basic Metals	2.05	9.01
28	Fabricated Metal Products	2.47	12.13
29	Machinery and Equipment	2.65	13.03
30	Office, Accounting & Computing Machinery	4.75	23.67
31	Electrical Machinery and Apparatus	4.75	21.57
32	Communication Equipment	4.62	19.00
33	Medical, Precision and Optical Instruments	2.82	7.42
34	Motor vehicles, Trailers and Semi-Trailers	0.39	1.28
35	Other transport equipment	1.51	20.74
36	Furniture; Manufacturing n.e.c	2.56	7.17
	Average	4.48	15.10

Table 15: Chinese Imports: By Industries - Before and After 2001

Notes: Numbers represent average across each industrial category according to National Industrial Classification (NIC) 2004 2-digit level. 'Chinese Imports/World Imports' is the share of Chinese imports in total imports of India. Source: Chakraborty and Henry (2018).

 Table 16: Outsourcing of Manufacturing Jobs - Total Expenditure, Share of Expenses, Percentage of Firms: User-based Industries

maabtrice			
Industry Name	Ν	Outsou Janufactur	rcing ring Jobs
	Total	Share	% of Firms
	(1)	(2)	(3)
Basic Goods	32.46	0.50	12.55
Intermediate	30.74	0.30	12.81
Capital Goods	46.16	0.29	12.86
Consumer Durables	36.50	0.77	18.51
Consumer Non-Durables	46.43	0.64	16.30

Notes: Numbers represent average across manufacturing firms belonging to each user-based industry. Column (1) calculates the mean outsourcing expenditure by an Indian manufacturing firm. It is expressed in INR Million. Column (2) represents the mean share of outsourcing expenditure in total expenditure of a firm multiplied by 100. Column (3) represents mean percentage of firms involved in outsourcing of manufacturing jobs.

Table 17: Im	port Co	npetitio	n and O ₁	tsourcing I	ntensity:	Testing	for the	Extensiv	re Margin	I	
					0	utsourcing Intensity					
					1995-2	r 007					$_{1995-2001}^{\mathrm{Year}}$
				$\underset{OutExp_{t-1}}{\operatorname{Control}}$	$\mathop{\mathrm{First}}_{\mathrm{Diff}}$	$\underset{\text{Diff}}{\text{Long}}$	FComp UEA	${\operatorname{LR}}_{(2013)}$	FLogit	Trend Break Analysis	
	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)	(6)	(10)	(11)
$DComp_{IN,jt-1}^{China}$ $(OutMan Jobs/TE)_{it-1}$	$0.165^{***}_{(0.052)}$	0.103^{**} $_{(0.059)}$	$0.115^{**}_{(0.058)}$	$\begin{array}{c} 0.131^{**} \\ (0.051) \\ 2.373^{***} \end{array}$	$0.184^{***}_{(0.064)}$	$\begin{array}{c} 0.034 \\ (0.047) \end{array}$	$0.132^{**}_{(0.057)}$	$\begin{array}{c} 4.242 \\ (4.004) \end{array}$	$0.239^{***}_{(0.032)}$		$0.968 \\ (0.996)$
$InpTariff_{jt-1}$		-0.071	-0.022	(0.276)	0.088^{***}	-0.653	-0.075	-0.071			-0.088
$Out Tariff_{jt-1}$		(0.016)	(0.04.0) -0.024 (0.032)		-0.044^{**}	(0.457) (0.457) (0.427)	(0.031)	(0.034) -0.013 (0.034)			(0.042)
$FComp_{IN,jt-1}^{China}$		0.002	0.002		0.012^{**}	0.006	0.004	0.002			0.001
$DComp_{IN,jt-1}^{Other\ LWC}$		-0.044	-0.051		(2000)	(2000)					
$DComp_{IN,jt-1}^{China} imes (t-2001) Trend$										-0.214^{**}	
$DComp_{IN,jt-1}^{China}\times(2002-2007)Trend$										0.346^{**}	
Firm $Controls_{t-1}$	$\mathbf{Y}_{\mathbf{es}}$	γ_{es}	γ_{es}	\mathbf{Yes}	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Y}_{\mathbf{es}}$	γ_{es}	Yes	Yes
R-Square	0.61	0.61	0.62	0.62	0.05	0.83	0.61	0.63	n/a	0.62	0.61
N E	41,579	39,233	33,448	41,821	38,072	7,147	39,723	38,676	46,163	41,515	17,841
Vear FE	Ves	Ves	Ves	Ves	ON ON	NO	Ves	Ves	Ves	1 CS Ves	1 CS Ves
Industry FE(4-digit)*Year Trend	Yes	$_{\rm Yes}$	$\mathbf{Y}^{\mathbf{es}}$	Yes	Yes	${ m Yes}$	Yes	${ m Yes}$	${\rm Yes}$	Yes	Yes
State FE*Year FE	No	No	$\mathbf{Y}_{\mathbf{es}}$	No	No	N_{O}	No	No	No	No	No
Notes: Columns $(1) - (12)$ use c	utsourcin	ng intensi	ty (it tak	es a value 1 v	when outso	ourcing e	xpenditu	re on ma	nufacturi	ng jobs of a	irm is
greater than zero) as the depend	lent vari	able. We	treat this	as the exten	sive margi	n of outs	ourcing.	$^{I}_{DComp_{I}^{C}}$	$_{N,jt-1}^{hina}$, is t	he Chinese i	mport
pentration ratio in the domestic n	narket of	India. It	is calcula	ted as the sh	are of Chi	nese imp	orts in ir	dustry j	at time t	by India div	rided by
total domestic production, impo	orts and	exports fo	or industr	y <i>j</i> in 1994 fc	or India.	InpTariff	, and O_u	tTariff'	are the na	tural logarit	am of
input and output tariffs faced by	Indian i	ndustries	at 2004 l	NIC 4-digit le	vel. 'FCon	$p_{IN,jt-1}^{China}$	is the m	easure of	Chinese	import comp	etition
faced by Indian firms in an e	xport de	stination	(US) exce	pt for colum	n (7) , whe	sre we inc	clude Chi	inese imp	orts by E	U and ASE/	N N
additionally. ' $DComp_{IN,jt-1}^{Other \ LWC}$ '	is the sh	are of im	ports fro	n all other lo	w-wage co	untries.	outMan	$ obs/TE_{it-} $	$_{1}$, is the l	agged depen	dent
variable. $(t - 2001)Trend'$ capture	s the diff	erntial pı	re-trend a	nd post-trene	d of Chine	se impor	t compet	ition afte	er China j	oined the W	$\Gamma O in$
2001; whereas ' $(2002 - 2007)Tren$	<i>i</i> ' only ce	ptures a	fixed time	e trend after	the Chine	se memb	ership of	the WT ⁰	O in 2001	. 'Firm Cont	rols'
include age, age squared of a firm,	size (ass	tets) and	technolog	y adoption (s	sum of R&	cD expen	diture ar	id Techno	ology Tra	nsfer). Both	`Assets'
and 'Technology Adoption' are	used at p	eriod $t - 1$	1 and in r	eal terms. St	andard Er	rors in p	arenthese	es are clu	istered at	the industry	level.

Intercepts are not reported. *,**,*** denotes 10%, 5% and 1% level of significance, respectively.

$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$											
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$			Outsom	rcing Exp	enses/			Ou	utsourcing		
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $			To	tal Expense	s				Intensity		
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$		$\substack{\text{Year}\\1992-2007}$	${ m Year}_{\leq 2003}$	${ m Year}_{\leq 2001}$	${ m Year}_{\leq 1999}$	${ m Year}_{\leq 1997}$	$_{ m 1992-2007}^{ m Year}$	${ m Year}_{\leq 2003}$	${ m Year}_{\leq 2001}$	${ m Year}_{\leq 1999}$	${ m Year}_{\leq 1997}$
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $		(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)	(6)	(10)
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	$InpTariff_{jt-1}$	-0.004 (0.003)	-0.006 (0.005)	-0.001 (0.003)	$\begin{array}{c} 0.0001 \\ (0.002) \end{array}$	$\begin{array}{c} 0.001 \\ (0.001) \end{array}$	-0.066 (0.047)	-0.107^{**} (0.052)	-0.038 (0.040)	-0.022 (0.031)	-0.036 (0.042)
Firm Controls_{t-1}Yes<	$OutTariff_{jt-1}$	0.003 (0.002)	0.004 (0.003)	0.002 (0.002)	$\begin{array}{c} 0.0002 \\ (0.001) \end{array}$	-0.0002 (0.001)	-0.008 (0.027)	$\begin{array}{c} 0.001 \\ (0.032) \end{array}$	-0.003 (0.028)	-0.018 (0.020)	-0.023 (0.029)
$ \begin{array}{llllllllllllllllllllllllllllllllllll$	Firm $Controls_{t-1}$	Yes	\mathbf{Yes}	Yes	Yes	Yes	Yes	\mathbf{Yes}	Yes	Yes	Yes
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	R-Square	0.58	0.55	0.54	0.58	0.69	0.59	0.56	0.56	0.62	0.63
Firm FEYes<	Z	56,281	36,211	28,080	20,073	13,250	56,281	36,211	28,080	20,073	13,250
Year FEYes<	Firm FE	Yes	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}
try FE (4-digit)*Year Trend Yes	Year FE	Yes	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}
s: Columns $(1) - (5)$ use expenditure on outsourcing of manufacturing jobs as a share of total expenses as the dependent variable. is $(6) - (10)$ use % of firms involved in outsourcing (it takes a value 1 when outsourcing expenditure on manufacturing jobs of a firm eater than zero) as the dependent variable. We treat the former as the intensive and latter as the extensive margin of outsourcing. ariff' and $OutTariff'$ are the natural logarithm of input and output tariffs faced by Indian industries at 2004 NIC 4-digit level. rm Controls' include age, age squared of a firm, size (assets) and technology adoption (sum of R&D expenditure and Technology rm Controls' include age, age squared of a firm, size (assets) and technology adoption (sum of R&D expenditure and Technology rm Controls' include age, age squared of a firm, size (assets) and technology adoption (sum of R&D expenditure and Technology rm Controls' include age, age squared of a firm, size (assets) and technology adoption (sum of R&D expenditure and Technology rm Controls' include age, age squared of a firm, size (assets) and technology adoption (sum of R&D expenditure and Technology rm Controls' include age, age squared of a firm, size (assets) and technology adoption (sum of R&D expenditure and Technology rm Controls' include age, age squared of a firm, size (assets) and technology adoption (sum of R&D expenditure and Technology rm Controls' include age, age squared of a firm, size (assets) and technology adoption (sum of R&D expenditure and Technology rm Controls' include age, age squared of a firm, size (assets) and technology adoption (sum of R&D expenditure and Technology rm Controls' include age, age squared of a firm, size (assets) are not reported. ** denotes 5% level of significance.	try FE (4-digit)*Year Trend	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}
is (6) – (10) use % of firms involved in outsourcing (it takes a value 1 when outsourcing expenditure on manufacturing jobs of a firm eater than zero) as the dependent variable. We treat the former as the intensive and latter as the extensive margin of outsourcing. ariff' and $OutTariff'$ are the natural logarithm of input and output tariffs faced by Indian industries at 2004 NIC 4-digit level. rm Controls' include age, age squared of a firm, size (assets) and technology adoption (sum of R&D expenditure and Technology r). Both 'Assets' and 'Technology Adoption' are used at period $t-1$ and in real terms. Standard errors in parentheses are clustered at the industry level. Intercepts are not reported. ** denotes 5% level of significance.	s: Columns $(1) - (5)$ use expend	iture on outs	sourcing of	f manufact	uring jobs	as a share	of total expe	enses as the	dependent	variable.	
eater than zero) as the dependent variable. We treat the former as the intensive and latter as the extensive margin of outsourcing. ariff and $OutTariff$ are the natural logarithm of input and output tariffs faced by Indian industries at 2004 NIC 4-digit level. rm Controls' include age, age squared of a firm, size (assets) and technology adoption (sum of R&D expenditure and Technology rm Controls' include age, age squared of a firm, size (assets) and technology adoption (sum of R&D expenditure and Technology rm Controls' include age, age squared of a firm, size (assets) and technology adoption (sum of R&D expenditure and Technology rm Controls' include age, age squared of a firm, size (assets) and technology adoption (sum of R&D expenditure and Technology rm Controls' include age, age squared of a firm, size (assets) and technology adoption (sum of R&D expenditure and Technology rm Controls' include age, age squared of a firm, size (assets) and technology adoption (sum of R&D expenditure and Technology rm Controls' include age, age squared of a firm, size (assets) and technology adoption (sum of R&D expenditure and Technology rm Controls' include age, age squared of a firm, size (assets) and technology adoption (sum of R&D expenditure and Technology rm Controls' include age, age squared of a firm, size (assets) and technology adoption (sum of R&D expenditure and Technology rm Controls' and 'Technology Adoption' are used at period $t-1$ and in real terms. Standard errors in parentheses are clustered at the industry level. Intercepts are not reported. ** denotes 5% level of significance.	is $(6) - (10)$ use % of firms invol	ved in outso	urcing (it	takes a val	ue 1 when	outsourcin	g expenditu	re on manuf	acturing jo	bs of a firm	_
ariff, and 'OutTariff' are the natural logarithm of input and output tariffs faced by Indian industries at 2004 NIC 4-digit level. rm Controls' include age, age squared of a firm, size (assets) and technology adoption (sum of R&D expenditure and Technology r). Both 'Assets' and 'Technology Adoption' are used at period $t - 1$ and in real terms. Standard errors in parentheses are clustered at the industry level. Intercepts are not reported. ^{**} denotes 5% level of significance.	sater than zero) as the dependen	t variable. V	Ve treat th	le former a	us the inter	isive and la	tter as the ϵ	extensive ma	rgin of out	sourcing.	
rm Controls' include age, age squared of a firm, size (assets) and technology adoption (sum of R&D expenditure and Technology r). Both 'Assets' and 'Technology Adoption' are used at period $t-1$ and in real terms. Standard errors in parentheses are clustered at the industry level. Intercepts are not reported. ^{**} denotes 5% level of significance.	ariff, and ' $OutTariff$ ' are th	le natural log	garithm of	input and	output ta	riffs faced l	oy Indian ind	dustries at 2	004 NIC 4-	digit level.	
r). Both 'Assets' and 'Technology Adoption' are used at period $t-1$ and in real terms. Standard errors in parentheses are clustered at the industry level. Intercepts are not reported. ^{**} denotes 5% level of significance.	rm Controls' include age, age squ	ared of a fir	m, size (as	ssets) and	technology	adoption ((sum of R&I) expenditur	re and Tech	nology	
at the industry level. Intercepts are not reported. ** denotes 5% level of significance.	r). Both 'Assets' and 'Technolog	y Adoption'	are used a	at period t	-1 and in	n real term	s. Standard	errors in pa	rentheses a	re clustered	_
	at the indus	try level. Int	cercepts ar	e not repo	rted. ^{**} de	motes 5% l	evel of signif	icance.			

Table 18: India's Trade Reforms and Outsourcing of Manufacturing Jobs

	0		· · · · · · · · · · · · · · · · · · ·	
	Outsour	cing Expe	nses (Manuf	acturing Jobs)/
		T	otal Expenses	
	Size	$\operatorname{End}_{\operatorname{Heo}}$	Export	Ownership
		Ose	Orientation	
	(1)	(2)	(3)	(4)
$DComp_{IN}^{China}_{it=1} \times Small \ Firm$	0.020^{**}			
	(0.010)			
$DComp_{IN,,jt-1}^{China} imes Big \ Firm$	0.026^{**}			
$DComp_{IN}^{China}_{it-1} imes Final$		0.020^{**}		
		(0.010)		
$DComp_{IN,,jt-1}^{China} imes Intermediate$		0.008^{**}		
$DComp_{IN}^{China}_{it=1} \times Exporter$		(0000)	-0.003	
$- T - 2 \int (t_{ATT} - t_{ATT}) dt_{ATT}$			(0.00)	
$DComp_{IN,jt-1}^{China} \times Non - Exporter$			0.014^{**}	
$DComp_{IN,it=1}^{China} imes Domestic$			(0000)	0.014^{***}
				(0.005)
$DComp_{IN,,jt-1}^{China} imes Foreign$				-0.006
Firm $Controls_{t-1}$	Yes	Yes	Yes	Yes
R-Square	0.56	0.56	0.56	0.56
Ν	41,515	41,515	41,515	41,515
Firm FE	Yes	\mathbf{Yes}	${ m Yes}$	${ m Yes}$
Year FE	Yes	\mathbf{Yes}	${ m Yes}$	${ m Yes}$
Industry FE (4-digit)*Year Trend	Y_{es}	$\mathbf{Y}_{\mathbf{es}}$	${ m Yes}$	\mathbf{Yes}

Table 19: Import Competition and Outsourcing of Manufacturing Jobs: Firm Characteristics

ndustry j at time t by India divided by total domestic production, imports and exports for industry j in 1994 for India. 'Small Firm' is $DComp_{IN,jt-1}^{China}$ is the Chinese import pentration ratio in the domestic market of India. It is calculated as the share of Chinese imports in produces consumer durable and non-durable products. *Intermediate*' is a dummy variable which takes a value 1 if the industry produces Foreign' takes a value 1 in case a firm has foreign ownership. All regressions control for $FComp_{IN,jt}^{China}$, and its interaction terms. Firm an indicator for firms belonging to the 1st and 2nd quartiles. 'Big Firm' is an indicator for firms belonging to the 3rd and 4th quartiles. Quartiles $(Qr_{i=1,2,3,4})$ are defined according to the total sales of a firm. 'Final' is a dummy variable that takes a value 1 if the industry basic, capital and intermediate goods. 'Exporter' is a variable which takes a value 1 if a firm exports. 'Non - Exporter' is a variable which assumes a value 1 if a firm's export flow is 0. 'Domestic' is a variable which assumes a value 1 if a firm's ownership is domestic. Both 'Assets' and 'Technology Adoption' are used at period t-1 and in real terms. Standard errors in parentheses are clustered at the Controls' include age, age squared of a firm, size (assets) and technology adoption (sum of R&D expenditure and Technology Transfer). Notes: Columns (1) $\overline{-}$ (4) use expenditure on outsourcing of manufacturing jobs as a share of total expenses as the dependent variable. industry level. Intercepts are not reported. *, **, *** denotes 10%, 5% and 1% level of significance, respectively.

ourcing	(7) (8)	-0.020 (0.038)	0.176^{***} (0.066)	-0.105 (0.116)	$\begin{array}{cccc} 0.207^{***} & -0.031 \\ (0.061) & (0.164) \end{array}$	Yes Yes	0.54 0.52	36,064 $36,064$	Yes Yes	Yes Yes	Yes Yes	d columns $(5) = (8)$ us
Outso	(9)			0.149^{***} (0.032)		\mathbf{Yes}	0.53	38,966	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	ne sesueux
	(5)	$0.069^{***}_{(0.011)}$				\mathbf{Yes}	0.51	41,579	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	o of total o
s/	(4)	-0.0007 (0.007)	$0.019^{**}_{(0.009)}$	-0.016 (0.017)	$\begin{array}{c} 0.003 \\ (0.051) \end{array}$	\mathbf{Yes}	0.63	36,064	${ m Yes}$	${ m Yes}$	${ m Yes}$	he ac a chai
Expenses	(3)				$\begin{array}{c} 0.033^{*} \\ (0.018) \end{array}$	$\mathbf{Y}_{\mathbf{es}}$	0.60	36,064	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Y}_{\mathbf{es}}$	of mining
utsourcing T ^{otal E}	(2)			$\begin{array}{c} 0.026^{***} \\ (0.008) \end{array}$		\mathbf{Yes}	0.58	38,966	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Y}_{\mathbf{es}}$	funnen for
Ō	(1)	$\begin{array}{c} 0.011^{***} \\ (0.003) \end{array}$				\mathbf{Yes}	0.56	41,821	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	Outeonioguo
		$DComp_{IN,,jt}^{China}$	$DComp_{IN,,jt-1}^{China}$	$DComp_{IN,,jt-2}^{China}$	$DComp_{IN,,jt-3}^{China}$	Firm $Controls_{t-1}$	R-Square	Ν	Firm FE	Year FE	ndustry FE (4-digit)*Year Trend	Columns (1) = (4) use expenditure on

fects
m Efl
g-terı
l Lon
- and
Short
Jobs:
uring
ufact
Man
JC
urcing e
Outso
and
npetition
μ
ő
mport Cc
0: Import Cc
e 20: Import Cc
able 20: Import Cc

of firms involved in outsourcing (it takes a value 1 when outsourcing expenditure on manufacturing jobs of a firm is greater than zero) as the dependent variable. We treat the former as the intensive and latter as the extensive margin of outsourcing. $DComp_{IN,jt}^{Chima}$, is the Chinese firm, size (assets) and technology adoption (sum of R&D expenditure and Technology Transfer). Both 'Assets' and 'Technology Adoption' import pentration ratio in the domestic market of India. It is calculated as the share of Chinese imports in industry j at time t by India divided by total domestic production, imports and exports for industry j in 1994 for India. 'Firm Controls' include age, age squared of a are used at period t-1 and in real terms. Standard errors in parentheses are clustered at the industry level. Intercepts are not reported. *, **, denotes 10%, 5% and 1% level of significance, respectively. Notes:

Regulations	States with neutral labour laws	(3)	Assam	Bihar	Haryana	Jammu and Kashmir	Punjab	Kerela	Madhya Pradesh
ates according to Labour	States with pro-employer labour laws	(2)	Andhra Pradesh	${ m Karnataka}$	${ m Rajasthan}$	Tamil Nadu	$\mathbf{U} \mathbf{t} \mathbf{t} \mathbf{a} \mathbf{r} \mathbf{r} \mathbf{a} \mathbf{d} \mathbf{e} \mathbf{h}$		
Table 21: St.	States with pro-worker labour laws	(1)	Gujarat	Maharashtra	Orissa	West Bengal			

Notes: Columns $(\overline{1}) - (\overline{3})$ list Indian states according to types of labour regulation according to \overline{G} upta et al. (2009).

Extensive	
sting for Labour Market Regulations –	
Manufacturing Jobs: Te	
1 Outsourcing of 1	
Competition and	
Import	
Table 22:	Margin

urcing	- Kuter	(2)	$0.095^{***}_{(0.017)}$	-0.068^{**} (0.029)	-0.018 (0.013)	-0.009 (0.016)	$\begin{array}{c} 0.003 \\ (0.010) \end{array}$	0.008 (0.014)	0.001^{**} (0.0004)	-0.001^{*}	-0.005	(0.013) (0.017)	\mathbf{Yes}	0.51	36,135	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	Yes
Outson		(1)	$0.116^{***}_{(0.015)}$	-0.103^{***} (0.023)	-0.021^{*}	-0.004 (0.016)	$0.004 \\ (0.009)$	0.007 (0.013)	~				${ m Yes}$	0.51	38, 131	$\mathbf{Y}^{\mathbf{es}}$	$\mathbf{Y}_{\mathbf{es}}$	\mathbf{Yes}	Yes
			$DComp^{China}_{IN,,jt-1}$	$DComp_{IN,,jt-1}^{China} imes LMktR_s$	$InpTariff_{jt-1}$	$InpTariff_{jt-1} imes LMktR_s$	$OutTariff_{jt-1}$	$OutTariff_{jt-1} imes LMktR_s$	$FComp_{IN,jt-1}^{China}$	$FComp_{IN,jt-1}^{Chima} \times LMktR_s$	$DComp_{IN,jt-1}^{Other\ LWC}$	$DComp_{IN,jt-1}^{Other \ LWC} imes \ LMktR_s$	Firm $Controls_{t-1}$	R-Square	Ν	Firm FE	Year FE	Industry FE (3-digit)*Year FE	State FE*Year FE

Indian industries at 2004 NIC 4-digit level. ' $FComp_{IN,jt-1}^{China}$ ' is a measure of Chinese import competition faced by Indian firms in an export Notes: Columns (1) - (2) use % of firms involved in outsourcing (it takes a value 1 when outsourcing expenditure on manufacturing jobs of squared of a firm, size (assets) and technology adoption (sum of R&D expenditure and Technology Transfer). Both 'Assets' and 'Technology Adoption' are used at period t-1 and in real terms. Standard errors in parentheses are clustered at the industry level. Intercepts are not a firm is greater than zero) as the dependent variable. $DComp_{IN,jt-1}^{China}$, is the Chinese import pentration ratio in the domestic market of India. It is calculated as the share of Chinese imports in industry j at time t by India divided by total domestic production, imports and exports for industry j in 1994 for India. InpTariff' and OutTariff' are the natural logarithm of input and output tariffs faced by destination (US). $DComp_{IN,jt-1}^{Other LWC}$, is the share of imports from all other low-wage countries. $LMktR_s$, is an indicator for labour market regulation. It takes a value 1 if a state has pro-employer labour market laws and 0 otherwise. Firm Controls' include age, age reported. *, **, aenotes 10%, 5% and 1% level of significance, respectively.

C Proof of Proposition

Proof. 1

Implicitly differentiating 13 at the optimal k_i^* with respect to σ

$$\Gamma\gamma_i^{-1}(c-w)p_i^{*-\sigma}\log(p_i^*) + \sigma\Gamma p_i^{*-\sigma-1}\gamma_i^{-1}(c-w)\frac{\delta p_i^*}{\delta\sigma} - G''(k_i^*)\frac{\delta k_i^*}{\delta\sigma} = 0$$
(15)

From 12

$$\frac{\delta p_i^*}{\delta \sigma} = -\gamma_i^{-1} (k_i c + (1 - k_i)w) \frac{1}{(\sigma - 1)^2} + \frac{\delta p_i^*}{\delta k_i^*} \frac{\delta k_i^*}{\delta \sigma}$$
(16)

Substituting into 15 and utilizing 14

$$\Gamma \gamma_i^{-1}(c-w) p_i^{*-\sigma} \log(p_i^*) - \sigma \Gamma p_i^{*-\sigma-1} \gamma_i^{-1}(c-w) \gamma_i^{-1}(k_i c + (1-k_i)w) \frac{1}{(\sigma-1)^2} + \frac{\delta^2 \pi_i}{\delta k_i^2} \frac{\delta k_i^*}{\delta \sigma} = 0$$

$$\frac{\delta k_i^*}{\delta \sigma} = -\frac{\Gamma \gamma_i^{-1}(c-w) p_i^{*-\sigma} \log(p_i^*) - \sigma \Gamma p_i^{*-\sigma-1} \gamma_i^{-1}(c-w) \gamma_i^{-1}(k_i c + (1-k_i)w) \frac{1}{(\sigma-1)^2}}{\frac{\delta^2 \pi_i}{\delta k_i^2}}$$

From 12 and given demand, this equals

$$\frac{\delta k_i^*}{\delta \sigma} = -\frac{\gamma_i^{-1} y_i^* (c-w) (\log(p_i^*) - \frac{1}{\sigma - 1})}{\frac{\delta^2 \pi_i}{\delta k_i^2}}$$

The sign of this derivative depends on the sign of the expression in brackets $(\log(p_i^*) - \frac{1}{\sigma-1})$. For small enough p_i , we can approximate this as

$$(p_i^*) - \frac{1}{\sigma - 1} = \frac{1}{\sigma - 1} \left(\frac{\sigma k_i^* c + (1 - k_i^*) w}{\gamma_i} - 1 \right)$$
(17)

Hence, as long as $\sigma k_i^* c + (1 - k_i^*) w < \gamma_i^*, \ \frac{\delta k_i^*}{\delta \sigma} > 0.$